Bubbles, Crashes, and Economic Growth: Theory and Evidence *

Pablo A. Guerron-Quintana[†] Tomohiro Hirano[‡] Ryo Jinnai[§]

June 1, 2022

Abstract

We analyze the ups and downs in economic growth in recent decades by constructing a model with recurrent bubbles, crashes, and endogenous growth. Once realized, bubbles crowd in investment and stimulate economic growth, but expectation about future bubbles crowds out investment and reduces economic growth. We identify bubbly episodes by estimating the model using the U.S. data. Counterfactual simulations suggest that the IT and housing bubbles not only caused economic booms but also lifted U.S. GDP by almost 2 percentage points permanently, but the economy could have grown even faster if people had believed that asset bubbles would never arise.

1 Introduction

A decade after the Great Recession, economic observers seem to agree on a few points. First, an asset price bubble emerged in the years leading up to the crisis. Second, economic growth was strong when the bubble was present. Third, the implosion of the bubble triggered a financial

^{*}We benefited from useful comments by Levent Altinoglu, Gadi Barlevy, Susanto Basu, Fernando Broner, Bernard Dumas, Andrew Foerster, Masashige Hamano, Takashi Kamihigashi, Michihiro Kandori, Nobuhiro Kiyotaki, Nan Li, Alberto Martin, Kiminori Matsuyama, Masaya Sakuragawa, Jose Scheinkman, Joseph Stiglitz, Dongho Song, Jean Tirole, Vincenzo Quadrini, Rosen Valchev, Jaume Ventura, and seminar participants at Aoyama Gakuin University, Bank of Canada, Bank of Japan, Beijing University, Boston College, Canon Institute for Global Studies, CREI, Emory, Espol, Hitotsubashi University, Japan Center for Economic Research, Keio University, Kobe University, Michigan State University, Monash University, National University of Singapore, Norwegian Business School, Okayama University, Osaka University, RIETI, Royal Holloway, University of London, Shanghai Jiao Tong University, Tokai University, Tohoku University, University of Birmingham, University of Tokyo, Waseda University, Wuhan University, and the NBER Summer Institute. Part of this work is supported by JSPS Kakenhi 18H00831, 18KK0361, 20H01490, and 20H00073.

[†]Boston College, pguerron@gmail.com

[‡]Royal Holloway, University of London and Center for Macroeconomics, London School of Economics and Canon Institute for Global Studies, tomohih@gmail.com

[§]Hitotsubashi University, rjinnai@ier.hit-u.ac.jp

crisis, resulting in severe contraction (Brunnermeier and Oehmke, 2013). Fourth, the recovery from the crisis was slower than the recovery from typical recessions. Recent empirical studies find that these observations are far from unique to the Great Recession or the U.S. Specifically, bubble-driven financial crises have happened in other countries; the emergence and the bust of asset bubbles were not extremely rare but repeated over time, with an interval of a few decades in many cases; and they were accompanied by ups and downs in economic growth (Kindleberger, 2001; Cerra and Saxena, 2008; Blanchard et al., 2015; Jorda et al., 2015). Motivated by these empirical findings, we construct a model with recurrent bubbles, crashes, and endogenous growth that can account for the aforementioned empirical regularities and that can be taken to the data for structural estimation.

In order to estimate periods of asset bubbles, we employ a regime-switching model and introduce two regimes: a "fundamental regime" and a "bubbly regime." The fundamental regime is characterized by the absence of bubbles (bubbleless economy), in which investors are unable to obtain funds as they wish because of financial frictions. An asset bubble may emerge when the economy switches to the bubbly regime. Under some conditions, we show that there exists an equilibrium in which asset bubbles emerge and collapse recurrently as the economy switches back and forth between the two regimes.

Theoretically, recurrent bubbles produce two competing effects in our framework. First, bubbles mitigate the investor's lack of funding problem once they appear, speeding up capital accumulation, which in our endogenous growth model speeds up economic growth. This is the so-called crowding-in effect of realized bubbles.¹ Second, there is a novel crowding-out effect generated by expectations of future bubbles. Households in our model are long-lived and experience the emergence and the collapse of bubbles recurrently. Importantly, they fully anticipate these dynamics. Therefore, even if bubbles are absent today, households expect their emergence in the future. Likewise, when bubbles exist, households rationally anticipate their future collapse and re-emergence. These expectations about future bubbles affect households' decisions and, crucially, are a drag on economic growth. The underlying mechanism is a wealth effect. That is, households will be wealthier when bubbles arise in the future. With this anticipation, households increase both consumption and leisure, which crowds out investment and reduces economic growth today.

We show that the crowding-out effect of future bubbles is quantitatively important. If bubbles do not appear frequently and the economy's financial market is severely under-developed, the standard crowding-in effect of realized bubbles could still dominate. On the other hand, if the financial market is relatively developed, the crowding-out effect of future bubbles can dominate. In this case, recurrent bubbles reduce average growth and welfare over the long run. Importantly, if bubbles emerge more frequently, the crowding-out effect becomes stronger. Therefore, high-

¹There is a crowding-out effect of realized bubbles too. But we do not emphasize it in our paper for two reasons. First, it is not new but has already been discussed in the literature; see Kocherlakota (2009), Farhi and Tirole (2012), and Hirano and Yanagawa (2017), for example. Second, it is quantitatively small in our model.

frequency bubbles can be undesirable even in financially under-developed economies.

Empirically, we estimate the model using U.S. data for the period 1984-2017. We identify bubbles by exploiting the model's robust predictions that both GDP growth and the stock-marketto-GDP ratio are simultaneously high when bubbles exist. Using these observables, we find that at least two bubbly episodes are very likely in our sample: the first one from around 1997 to 2001, and the second one from around 2006 to the onset of the Great Recession. Both the asset market and GDP growth were strong in these periods, which our model attributes to the emergence of bubbles. But not all the booms are estimated to be bubbly. For example, our model attributes the strong GDP growth in the mid-1990s to favorable productivity shocks, for the stock market was not strong enough to justify the existence of bubbles. By the same token, a strong stock market alone does not necessarily imply a bubble in our estimation. This is the case in 2014 when the stock market was booming but growth was relatively weak.

A counterfactual simulation reveals that the U.S. economy significantly benefited from the realized bubbles for two reasons. First, it directly enjoyed bubble-driven economic booms. Second, investment booms during the bubbly episodes permanently raised the output level even after the bubbles had gone. We estimate that the two bubbly episodes combined permanently raised the level of U.S. GDP by about 2 percentage points. However, another counterfactual simulation suggests that the U.S. economy could have grown even faster. That is, if the economy were in a different equilibrium in which bubbles never arose and were never expected to emerge, GDP growth would be higher than the actual on average. This is because the crowding-out effect of future bubbles is absent.

The rest of the paper proceeds as follows. After highlighting the contributions of our paper to the existing literature, we describe the baseline model in Section 2. Section 3 provides both analytical solutions to a special case and intuitions. In Section 4, we calibrate the baseline model for quantitative exercises in subsequent sections. Section 5 discusses both the crowding-in and the crowding-out effects of recurrent bubbles. Section 6 discusses empirical findings. Section 7 concludes.

Related Work in the Literature

This paper is related to the seminal work on asset price bubbles in an infinite horizon economy, e.g., Bewley (1980), Scheinkman and Weiss (1986), Woodford (1990), Kocherlakota (1992, 2009), and Kiyotaki and Moore (2012). They consider either deterministic bubbles, which are expected to survive forever, or the stochastic bubbles developed by Weil (1987), which are expected to collapse but once they do, their re-emergence is not expected at all.² In contrast, we consider

 $^{^{2}}$ The same applies to the landmark papers on rational bubbles in an overlapping generations model; e.g., Samuelson (1958), Shell et al. (1969, Section 3), Townsend (1980), Tirole (1985), Diba and Grossman (1988), Farhi and Tirole (2012), and Martin and Ventura (2012). Martin and Ventura consider recurrent bubbly episodes, but

recurrent bubbly episodes in an infinite horizon economy. Infinitely lived households rationally expect that bubbles will repeatedly emerge and collapse in the future. These expectations about future bubbles will affect households' decisions, even when bubbles are absent today.³

Our work also relates to the recent papers highlighting the downside of asset bubbles. This is interesting research because, as Barlevy (2018) articulates, there is a concern that the theoretical literature on bubbles traditionally emphasizes their upside disproportionately and as a result does not address the types of issues policymakers care most about.⁴ Specifically, Allen et al. (2021) and Biswas et al. (2020) show that stagnation in output occurs after the bursting of bubbles in models without growth. The stagnation in output makes the welfare impact of bubbles nontrivial even if bubbles raise the output level when they are present. Their arguments, however, hinge on mechanisms that are not necessarily related to bubbles; namely, Allen et al. (2021) introduce default costs exogenously associated with the collapse of bubbles, and Biswas et al. (2020) introduce downward nominal wage rigidities. Our model abstracts from such mechanisms or other frictions, including nominal price rigidities or fire-sale externalities. Nonetheless, an interesting cost still emerges endogenously, just because infinitely lived households anticipate future bubbles.

Our paper is also related to the news shock literature (Beaudry and Portier, 2006; Jaimovich and Rebelo, 2009; Schmitt-Grohe and Uribe, 2012) and a recent paper by Schaal and Taschereau-Dumouchel (2021), in which expectations play a key role in generating business cycles. There are two crucial differences from our paper. First, in the news shock literature, expectations about future fundamentals, typically the productivity level, are the key driver. In contrast, expectations about nonfundamentals, i.e., asset bubbles, are important in our model. Second, in the news shock literature, both positive news and negative news are equally likely. Because their unconditional means are zero, the presence of the news shocks has no impact on the model's deterministic steady state. In contrast, economic agents in our model rationally anticipate how bubbles will evolve from

agents in their model live for only two periods, and in addition, everyone supplies one unit of labor inelastically in the young period and consumes only in the old period. These assumptions make expectations about future bubbles irrelevant to labor supply, consumption, and investment in the young period, as well as to welfare. It does not matter whether the emergence of future bubbles is expected or not. In this sense, their bubble model is essentially the same as Weil's stochastic-bubble model.

³Following Weil (1987), we assume that bubbles will collapse all at once when it comes to a crash moment. This assumption is strong if it is taken literally, but is convenient to capture large nonlinear effects of the crash. For example, in the 1980s asset price bubble in Japan, stock prices rose by more than 3 times between 1985 and 1989, and the urban land price rose by almost 4 times between 1985 and 1990; then, stock prices fell by 60 percent by 1992, and the urban land price fell by 80 percent by 1999 (Okina et al., 2001). Such large asset price movements will have large nonlinear effects on the economy, a point emphasized by studies on financial crises such as Mendoza (2010), Brunnermeier and Sannikov (2014), He and Krishnamurthy (2013), Gertler and Kiyotaki (2015), and Gertler et al. (2020). In order to capture large nonlinear effects, we assume complete collapses as in Weil (1987). A similar assumption is made by Gertler and Kiyotaki (2015) and Gertler et al. (2020); a bank run in their models is the entire collapse of the banking sector, with which they capture large nonlinear macroeconomic effects of the run. See Gali (2014) and Miao et al. (2015) for a local analysis around the bubbly steady state with a standard linearization method. The local analysis may be suitable to analyze the bubble economy without a large crash. But it would not be suitable to analyze the economy experiencing large crashes, for their effects would be nonlinear.

⁴The classic argument is that bubbles improve welfare because they help consumption smoothing (Samuelson, 1958; Bewley, 1980; Scheinkman and Weiss, 1986; Farhi and Tirole, 2012; Hirano and Yanagawa, 2017).

now on; in the fundamental regime, they expect a new bubbly episode to begin, and in the bubbly regime, they anticipate the crash. These regime-dependent expectations about asset bubbles have significant impacts on the regime-dependent steady states.⁵

Our empirical contribution is related to the growing body of work on quantitative models with endogenous growth. Comin and Gertler (2006) is a seminal contribution; they propose a model with endogenous productivity that can account for medium cycle properties of the U.S. data. More specifically, they demonstrate that an otherwise standard macroeconomic model with both research and development (R&D) and adoption stages reproduces the procyclical movements in technological change and R&D, while delivering reasonable business cycles. Guerron-Quintana and Jinnai (2019) examine the causes of the post-war U.S. recessions through the lens of an estimated dynamic stochastic general equilibrium model with both financial frictions and endogenous growth. In their framework, adverse financial shocks account for the severity of the Great Recession, but the same shocks could not account for why GDP growth slowed down after the crisis by about 1 percentage point. The reason is simple: the data counterpart of the financial shocks lacks persistence (Stock and Watson, 2012), and many financial indicators temporarily deteriorated but recovered after the bankruptcy of Lehman Brothers (Guerron-Quintana and Jinnai, 2019). In contrast, the current paper accounts for the growth slowdown intuitively. Namely, to the extent that the high growth in the 2000s was boosted by asset bubbles, the collapse of these bubbles led inevitably to slower growth. Furthermore, a new bubbly episode has not occurred vet according to our estimates.⁶

In a similar line of thinking, Anzoategui et al. (2019) propose and estimate an endogenous growth model with nominal frictions. The authors find that the slowdown in productivity in the 2010s resulted from a contraction in demand. This decrease in consumption triggered a reduction in R&D and hence the decline in productivity.⁷ Anzoategui et al. (2019), Comin and Gertler (2006), and Guerron-Quintana and Jinnai (2019) reach their conclusions based on models linearized around a unique steady state. As we show in the next sections, our model is more complex due to regime switching that affects steady states and growth rates. Morevoer, the solution and estimation techniques adopted in this paper are novel in the endogenous growth literature and bubble literature. The regime-switching feature in turn relates our contribution to the work on the solution and estimation of models with Markov regimes extensively reviewed by Hamilton (2016).

⁵Computing regime-dependent steady states is difficult in general. But in our model, endogenous growth simplifies the task. Namely, once we detrend the model using the endogenous state variable, the equilibrium conditions depend only on the exogenous state variables. Then, we can find the regime-dependent steady states of the detrended variables easily. See Section C.3 in the appendix for details.

⁶Hysteresis is another literature studying the growth slowdown. Blanchard et al. (2015) and Jorda et al. (2015) document this phenomenon not only for the U.S. but for other countries. Gali (2016) studies hysteresis in labor markets and the design of monetary policy. We view our work as complementary to this literature.

⁷Mian and Sufi (2015) argue that an elevated household debt was important for the contraction in demand. They discuss a bubble in the housing market, but they do so in a behavioral finance framework.

2 Model

Our model description consists of regimes, firms, and households.

2.1 Regimes

Let $z_t \in \{b, f\}$ denote a realization of the regime, where b and f denote the bubbly and fundamental regimes, respectively. Their defining characteristics are the existence or lack of bubbly assets. They are intrinsically useless, contributing neither to production nor to households' utility directly. In the fundamental regime, there are no bubbly assets in the economy. When the regime switches to a bubbly one, M units of (a new vintage of) bubbly assets are created and given to households in a lump-sum way. There is no bubble creation in other contingencies. Bubbly assets last without depreciation as long as the economy stays in the bubbly regime. They physically disappear at once when the regime switches back to the fundamental one.^{8,9} We assume that z_t follows a Markov process satisfying

$$\Pr(z_t = f | z_{t-1} = f) = 1 - \sigma_f$$

and

$$\Pr(z_t = b | z_{t-1} = b) = 1 - \sigma_b.$$

Here, σ_f and σ_b are bounded between 0 and 1.

2.2 Firms

Competitive firms produce output from capital and labor services denoted by KS_t^D and L_t^D , respectively. The production function is

$$Y_t = A_t \left(K S_t^D \right)^{\alpha} \left(L_t^D \right)^{1-\alpha},$$

where A_t is the technology level that agents in the economy take as given. Firms maximize profits defined as $Y_t - r_t K S_t^D - w_t L_t^D$ by choosing $K S_t^D$ and L_t^D , where r_t is the rental price of capital and w_t is the wage rate. The production technology is freely available to potential entrants. Firms make zero profits in equilibrium.

We assume that, while all the economic agents in the model take it as given, the technology

⁸Alternatively, we can assume that the prices of the bubbly assets become zero all at once. They are isomorphic.

⁹We assume complete collapses in order to capture the large nonlinear effects associated with the bursting of a bubble. In Section F in the appendix, we examine an alternative assumption, replacing the fundamental regime with a low-bubble regime in which a small fraction of bubbly assets survive from the previous regime. This model behaves quite differently from our baseline model, because when the economy moves to the low-bubble regime, demand for the surviving bubbly assets shoots up, pushing up their prices and hence mitigating the impact of the crash. The details are in the appendix.

level A_t is actually endogenous:

$$A_t = \bar{A} \left(K_t \right)^{1-\alpha} e^{a_t}.$$

Here, a_t is an exogenous productivity shock and \overline{A} is a scale parameter. Following Arrow (1962), Sheshinski (1967), and Romer (1986), we interpret the dependency of A_t on K_t as learning-bydoing; namely, knowledge is a by-product of investment, and in addition, it is a public good that anyone can access at zero cost. With this assumption, the long-run tendency for capital to experience diminishing returns is eliminated. Long-run growth is sustained by both capital and knowledge accumulation. Moreover, the growth rate is endogenous and influenced by not only the state of the economy but also actions taken by economic agents. This implication is crucial for our study, because we are interested in factors causing ups and downs in economic growth.

2.3 Households

The economy is populated by a continuum of households, with measure one. All households behave identically. Each household has a unit measure of members who are identical at the beginning of each period. During the period, members are separated from each other, and each member receives a shock that determines her role in the period. A member will be an investor with probability $\pi \in [0, 1]$ and will be a saver/worker with probability $1-\pi$. These shocks are i.i.d. among members and across time.

A period is divided into four stages: household's decisions, production, investment, and consumption. In the household's decision stage, all members of a household are together and pool their assets: n_t units of capital and \tilde{m}_t units of bubbly assets. Aggregate shocks are realized. The head of the household decides the capacity utilization rate u_t , i.e., how intensively to use the capital it owns. Because all members of the household are identical in this stage, the household head evenly divides the assets among the members. The household head also gives contingency plans to each member as follows. If the member becomes an investor, he or she spends i_t units of final goods to invest, and brings home the following items before the consumption stage: x_t^i units of final goods, n_{t+1}^i units of capital, and \tilde{m}_{t+1}^i units of bubbly assets. Likewise, if the member becomes a saver, he or she supplies l_t units of labor, and brings home the following items before the consumption stage: x_t^s units of final goods, n_{t+1}^s units of capital, and \tilde{m}_{t+1}^s units of capital, and \tilde{m}_{t+1}^s units of capital, and \tilde{m}_{t+1}^s units of bubbly assets. After receiving these instructions, members go to the market and remain separated from each other until the consumption stage.

At the beginning of the second stage, each member receives the shock that determines her role in the period. Markets open and competitive firms produce final goods. Compensation for productive factors is paid to their owners. A fraction of capital depreciates. Following Greenwood et al. (1988), we assume that a higher utilization rate causes a faster depreciation of the capital stock either because wear and tear increase with use or because less time can be devoted to maintenance. Specifically, the depreciation rate $\delta(u_t)$ is given by

$$\delta\left(u_{t}\right) = \delta_{0} + \frac{\delta_{1}}{1+\zeta}u_{t}^{1+\zeta}$$

where $\zeta > 0$.

Investors seek finance to undertake investment projects in the third stage. Financing comes through different channels: own resources, selling of new and existing capital, and, if in the bubbly regime, selling of bubbly assets. Investors have access to a linear technology that transforms any amount of final goods into the same amount of new capital. Asset markets close at the end of this stage. Members of the household meet again in the consumption stage. An investor consumes c_t^i units of final goods and a saver consumes c_t^s units of final goods. After consumption, members' identities are lost. They start a new period as identical members.

The instructions must meet a set of constraints. First, they have to satisfy an intratemporal budget constraint. For an investor, it is

$$x_{t}^{i} + i_{t} + \underbrace{q_{t}\left(n_{t+1}^{i} - i_{t} - \left(1 - \delta\left(u_{t}\right)\right)n_{t}\right)}_{\text{net capital purchase}} + \underbrace{\mathbf{1}_{\{z_{t}=b\}}\tilde{p}_{t}\left(\tilde{m}_{t+1}^{i} - \tilde{m}_{t}\right)}_{\text{net bubble purchase}} = u_{t}r_{t}n_{t}, \tag{1}$$

and for a saver, it is

$$x_{t}^{s} + q_{t} \left(n_{t+1}^{s} - (1 - \delta(u_{t})) n_{t} \right) + \mathbf{1}_{\{z_{t}=b\}} \tilde{p}_{t} \left(\tilde{m}_{t+1}^{s} - \tilde{m}_{t} \right) = u_{t} r_{t} n_{t} + w_{t} l_{t}.$$
(2)

Here, q_t and \tilde{p}_t denote prices of capital and bubbly assets, respectively. **1** is an indicator function to be discussed momentarily. In addition, the instructions must satisfy a feasibility constraint in the consumption stage given by

$$\pi x_t^i + (1 - \pi) x_t^s = \pi c_t^i + (1 - \pi) c_t^s.$$
(3)

In the fundamental regime, there are neither spot nor future markets for bubbly assets, and in addition, future bubbles cannot be used as collateral for loans.¹⁰ The indicator function in the budget constraints (1) and (2) captures this idea; the bubbly assets are worthless in the fundamental regime, because they are intrinsically useless and no one can sell them in the fundamental regime. We also impose the following restrictions:

$$\mathbf{1}_{\{z_t=f\}}\tilde{m}_{t+1}^i = \mathbf{1}_{\{z_t=f\}}\tilde{m}_{t+1}^s = 0,$$
(4)

¹⁰Tirole (1985) made an identical assumption regarding rent creation. He wrote: "An important feature of rent creation is that most rents are not capitalized before their "creation." For example a painting to be created by a 21st century master cannot be sold in advance by the painter's forebears. Similarly patents cannot be granted for future inventions." Our assumption regarding bubble creation is the same; we assume that bubble creation is exogenous, and bubbles cannot be capitalized before their creation.

meaning that no one can purchase bubbly assets in the fundamental regime because there are no markets for bubbly assets.

Following Kiyotaki and Moore (2012), we assume that an investor can issue new equity on, at most, a fraction ϕ of investment. In addition, she can sell, at most, a fraction ϕ of existing capital in the market too. Effectively, these constraints introduce a lower bound to the capital holdings at the end of the period:

$$n_{t+1}^{i} \ge (1 - \phi) \left(i_{t} + (1 - \delta \left(u_{t} \right)) n_{t} \right).$$
(5)

A similar constraint applies to n_{t+1}^s , but we omit it because it does not bind in equilibrium. We also omit non-negativity constraints for u_t , c_t^i , i_t , n_{t+1}^i , x_t^s , c_t^s , l_t , n_{t+1}^s , and \tilde{m}_{t+1}^s for the same reason. However, there are two exceptions

$$\tilde{m}_{t+1}^i \ge 0 \tag{6}$$

and

$$x_t^i \ge 0,\tag{7}$$

which mean that the investor can't short sell bubbly assets and must bring a non-negative amount of consumption back to the household.

The household's problem is summarized as follows. A sequence of u_t , x_t^i , c_t^i , i_t , n_{t+1}^i , \tilde{m}_{t+1}^i , x_t^s , c_t^s , l_t , n_{t+1}^s , and \tilde{m}_{t+1}^s is chosen to maximize the utility

$$E_0\left[\sum_{t=0}^{\infty} \frac{\beta^t}{e^{d_t}} \left(\pi \log\left(c_t^i\right) + (1-\pi)\left[\log\left(c_t^s\right) + \eta\left(1-l_t\right)\right]\right)\right]$$
(8)

subject to (1), (2), (3), (4), (5), (6), (7), and the laws of motion for assets given by

$$n_{t+1} = \pi n_{t+1}^i + (1 - \pi) n_{t+1}^s \tag{9}$$

and

$$\tilde{m}_{t+1} = \pi \tilde{m}_{t+1}^{i} + (1 - \pi) \tilde{m}_{t+1}^{s} + \mathbf{1}_{\{z_t = f, z_{t+1} = b\}} M$$

for all $t \ge 0$. d_t is a preference shock temporarily influencing the household's patience. It is a stylized demand-side shock to the economy. In addition, its effects resemble the effects of shocks originating from the financial sector in many ways (Fisher, 2014; Smets and Wouters, 2007). The initial portfolio is $\{n_0, \tilde{m}_0\} = \{K_0, \mathbf{1}_{\{z_t=b\}}M\}$ where K_t denotes the capital stock in the economy in period t.

2.4 Market Clearing

Competitive equilibrium is defined in a standard way; all agents optimize given prices, and the market clearing conditions are satisfied for capital

$$n_{t+1} = K_{t+1},$$

labor services

$$L_t^D = (1 - \pi) l_t,$$

capital services

$$KS_t^D = u_t K_t$$

and final goods

$$\pi c_t^i + (1 - \pi) c_t^s + \pi i_t = Y_t$$

for all t. If it is the bubbly regime $(z_t = b)$, the market clearing condition for the bubbly assets

$$\pi \tilde{m}_{t+1}^i + (1-\pi) \tilde{m}_{t+1}^s = M$$

is also satisfied. The law of motion for aggregate capital stock is

$$K_{t+1} = (1 - \delta(u_t)) K_t + \pi i_t,$$

which automatically holds by Walras' law.

2.5 Solving the Household's Problem

The financial constraint (5) does not bind if ϕ is sufficiently large. In this case, the price of capital is equal to one, and the price of bubbly assets is equal to zero, as we show in Section A.1 in the appendix. The household's problem becomes standard; it chooses a sequence of u_t , c_t^i , c_t^s , l_t , and n_{t+1} to maximize the utility subject to

$$\pi c_t^i + (1 - \pi) c_t^s + n_{t+1} = \left[u_t r_t + (1 - \delta (u_t)) \right] n_t + w_t (1 - \pi) l_t.$$

The first-order conditions are also standard. We show them in Section A.1 in the appendix.

If ϕ is small, the financial constraint (5) binds. The price of capital exceeds one, because capital is not only used as a production factor but also provides liquidity to its owners. Because capital creation is profitable, investors will increase i_t as much as possible, implying that the following feasibility constraint for investment holds in equilibrium:

$$\underbrace{(1 - \phi q_t) i_t}_{\text{minimum cost to conduct } i_t} = \underbrace{u_t r_t n_t + \phi q_t \left(1 - \delta \left(u_t\right)\right) n_t + \mathbf{1}_{\{z_t = b\}} \tilde{p}_t \tilde{m}_t}_{\text{maximum liquidity an investor can attain}}.$$
(10)

We derive this equation in Section A.2 in the appendix. The left-hand side is the minimum cost investors have to finance in order to conduct i_t , which is smaller than i_t because a part of the costs can be covered by selling newly created capital. The right-hand side is the maximum liquidity an investor can attain.

Combining (1), (2), (3), (9), and $\mathbf{1}_{\{z_t=b\}}\tilde{p}_t\tilde{m}_{t+1}^i = 0$, we obtain the budget constraint at the household level:

$$\pi c_t^i + (1 - \pi) c_t^s + \pi i_t + q_t [n_{t+1} - (1 - \delta (u_t)) n_t] + \mathbf{1}_{\{z_t = b\}} \tilde{p}_t \left[(1 - \pi) \tilde{m}_{t+1}^s - \tilde{m}_t \right]$$

= $u_t r_t n_t + \pi q_t i_t + (1 - \pi) w_t l_t.$ (11)

Substituting (10) into (11), we obtain

$$\pi c_{t}^{i} + (1 - \pi) c_{t}^{s} + q_{t} n_{t+1} + \mathbf{1}_{\{z_{t}=b\}} \tilde{p}_{t} (1 - \pi) \tilde{m}_{t+1}^{s}$$

$$= u_{t} r_{t} n_{t} + (1 - \pi) w_{t} l_{t} + (1 - \delta (u_{t})) q_{t} n_{t} + \mathbf{1}_{\{z_{t}=b\}} \tilde{p}_{t} \tilde{m}_{t}$$

$$+ \lambda_{t} \pi \underbrace{\left[u_{t} r_{t} n_{t} + \phi q_{t} (1 - \delta (u_{t})) n_{t} + \mathbf{1}_{\{z_{t}=b\}} \tilde{p}_{t} \tilde{m}_{t}\right]}_{\mathbf{y}},$$

$$(12)$$

maximum liquidity an investor can attain

where

$$\lambda_t \equiv \frac{q_t - 1}{1 - \phi q_t}.$$

This is an important equation. The left-hand side is gross spending, consisting of consumption and gross asset purchases. The first line in the right-hand side is gross income, consisting of dividends, labor income, and the market value of the portfolio. The second line in the right-hand side is the total profit from capital creation. The reason is the following. An investor can create $1/(1 - \phi q_t)$ units of capital from a unit of liquidity. A fraction ϕ of the investment is sold, and the rest is added to the investor's portfolio, which is worth $(1 - \phi) q_t/(1 - \phi q_t)$. Finally, subtracting the costs of the investment from it, we find

$$\frac{\left(1-\phi\right)q_{t}}{1-\phi q_{t}}-1=\frac{q_{t}-1}{1-\phi q_{t}}=\lambda_{t}$$

Hence, λ_t measures how much value an investor can create from a unit of liquidity. Finally, because investors as a group have $\pi \left[u_t r_t n_t + \phi q_t \left(1 - \delta \left(u_t \right) \right) n_t + \mathbf{1}_{\{z_t=b\}} \tilde{p}_t \tilde{m}_t \right]$ units of liquidity, the second line in the right-hand side is the total profit from capital creation at the household level.

The household's problem is now simplified. It chooses a sequence of u_t , c_t^i , c_t^s , l_t , n_{t+1} , and

 \tilde{m}_{t+1}^s to maximize the utility (8) subject to the budget constraint (12), the law of motion of bubbly assets

$$\tilde{m}_{t+1} = (1 - \pi) \, \tilde{m}_{t+1}^s + \mathbf{1}_{\{z_t = f, z_{t+1} = b\}} M,$$

and the absence of the bubbly-asset market in the fundamental regime

$$\mathbf{1}_{\{z_t=f\}}\tilde{m}_{t+1}^s = 0.$$

The first-order conditions are

$$c_{t}^{i} = c_{t}^{s},$$

$$\eta \frac{c_{t}^{s}}{1 - l_{t}} = w_{t},$$

$$r_{t} - \delta'(u_{t}) q_{t} + \pi \lambda_{t} (r_{t} - \phi q_{t} \delta'(u_{t})) = 0,$$

$$q_{t} = E_{t} \left[\frac{\beta}{e^{d_{t+1} - d_{t}}} \left(\frac{c_{t}^{s}}{c_{t+1}^{s}} \right) (u_{t+1}r_{t+1} + (1 - \delta(u_{t+1})) q_{t+1} + \pi \lambda_{t+1} (u_{t+1}r_{t+1} + \phi q_{t+1} (1 - \delta(u_{t+1})))) \right],$$
(13)

and

$$\mathbf{1}_{\{z_t=b\}}\tilde{p}_t = \mathbf{1}_{\{z_t=b\}}E_t \left[\frac{\beta}{e^{d_{t+1}-d_t}} \left(\frac{c_t^s}{c_{t+1}^s}\right) \left(1+\pi\lambda_{t+1}\right)\tilde{p}_{t+1}\mathbf{1}_{\{z_{t+1}=b\}}\right].$$
(14)

The first equation states that the marginal utility from consumption has to be equalized across members of the household. The second equation states that the marginal rate of substitution between leisure and consumption has to be equal to the wage. The third equation states that the marginal benefit of raising the capacity utilization rate (the rental rate of capital) has to be equal to its opportunity cost (the value of the depreciated capital at the margin). The fourth equation is the Euler equation for capital, in which λ_t appears because capital is not only a production factor but also a means of providing liquidity to investors. The fifth equation is the Euler equation for the bubbly assets. The left-hand side of this equation is strictly positive only if there is a chance that the bubbly assets in period t will be traded at a strictly positive price in the next period. In other words, it is the resalability of bubbly assets that justifies their positive prices.

On the optimal feasible plan, the transversality conditions must be satisfied too, which are given by

$$\lim_{t \to \infty} E_0 \left[\frac{\beta^t}{e^{d_t}} \left(\frac{1}{c_t^s} \right) q_t K_{t+1} \right] = 0$$

for the capital stock and

$$\lim_{t \to \infty} \mathbf{1}_{\{z_0=b\}} E_0 \left[\frac{\beta^t}{e^{d_t}} \left(\frac{1}{c_t^s} \right) \tilde{p}_t \mathbf{1}_{\{z_t=b\}} M \right] = 0$$

for bubbly assets. We show that they are satisfied along the balanced growth path.¹¹ See Section

¹¹Kocherlakota (1992) explicitly derived the economic conditions for which asset bubbles can arise in infinite

E in the appendix.

3 Analytical Solutions in a Simplified Model

Before the quantitative analysis of the full model, we study a special case in which analytical solutions are available. With the analytical solutions, we can see what the crowding-in effect of realized bubbles and the crowding-out effect of future bubbles are, and how they materialize in our setting. To this end, we make the following simplifying assumptions. First, the capital share α is equal to one. Second, the capacity utilization rate u_t is fixed at one. Third, capital fully depreciates every period. Fourth, both productivity and preference shocks are turned off.

Finally, we consider only one bubbly episode. That is, we assume that the economy initially stays in the fundamental regime, from which the economy will transition to the bubbly regime at a positive probability σ_f and people correctly recognize this possibility. Once the bubble emerges, it may persist for a while, and then may burst with a positive probability σ_b . After the bursting, no bubbles arise, i.e., the economy will stay in the fundamental regime forever. People rationally anticipate it too.

3.1 Capital Growth

We derive the growth rate of capital accumulation before, during, and after the bubbly episode first—derivations are presented in detail in Section D in the appendix. Let's start with the economy after the bubbly episode. The Euler equation for capital, equation (13) above, is written as

$$q_f^* = \beta \frac{r}{g_f^*} \left(1 + \pi \frac{q_f^* - 1}{1 - \phi q_f^*} \right), \tag{15}$$

where g_f^* is the growth rate of capital $(g_t \equiv K_{t+1}/K_t)$ and r is the rental rate, which is constant in this simplified model. The subscript "f" indicates that the variable is determined in the fundamental regime, and the asterisk indicates that it is determined in the fundamental regime after the bubbly episode but not before. Because both $K_{t+1} = \pi i_t$ and $n_t = K_t$ hold, the feasibility constraint for investment, equation (10), is written as

$$(1 - \phi q_f^*) g_f^* = \pi r.$$
 (16)

horizon economies in a manner fully consistent with the transversality condition. He showed in an endowment economy that if bubbles exist at any date, then everyone faces borrowing constraints that bind currently or at some point in the future. Furthermore, the economy needs to grow as fast as bubbles, so that agents can always afford to buy the bubbly asset. As in Kocherlakota (1992), in our model, on the balanced growth path with positive bubbles, the liquidity constraint and the short-sale constraint bind, and the economy grows at the same rate as the bubbles. The transversality condition is satisfied. See Kocherlakota (1992) for a more general discussion.

We solve equations (15) and (16) for q_f^* and g_f^* , obtaining

$$q_f^* = \frac{\beta \left(1 - \pi\right)}{\pi \left(1 - \beta\right) + \beta \phi} \tag{17}$$

and

$$g_f^* = r \left[1 - \frac{(1-\beta)(1-\pi)}{1-\beta+\beta\phi} \right].$$

The price of capital decreases with ϕ , because capital is less valuable as a source of liquidity if the financial constraint is loose. The growth rate of capital increases with ϕ , because investors obtain more liquidity.

Next, we consider the economy during the bubbly episode. The Euler equation for capital is written as

$$q_b = \frac{r}{g_b} \left[\left(1 - \sigma_b\right) \beta \left(1 + \pi \frac{q_b - 1}{1 - \phi q_b}\right) + \sigma_b \beta \left(\frac{\hat{c}_b}{\hat{c}_f^*}\right) \left(1 + \pi \frac{q_f^* - 1}{1 - \phi q_f^*}\right) \right],\tag{18}$$

where the subscript "b" indicates that the variable is determined in the bubbly regime, and \hat{c}_b and \hat{c}_f^* are aggregate consumption relative to the capital stock ($c_t \equiv Y_t - \pi i_t$ and $\hat{c}_t \equiv c_t/K_t$) in the bubbly regime and in the fundamental regime after the bubbly episode, respectively. The feasibility constraint for investment is written as

$$(1 - \phi q_b) g_b = \pi (r + m_b), \qquad (19)$$

where m_b is the size of the bubble relative to the capital stock $(m_t \equiv \tilde{p}_t \mathbf{1}_{\{z_t=b\}} M/K_t)$ in the bubbly regime. Finally, the Euler equation for the bubbly assets, equation (14), is written as

$$m_{b} = \underbrace{\left(1 - \sigma_{b}\right)}_{\text{bubble's survival probability}} \beta \left(1 + \pi \underbrace{\frac{q_{b} - 1}{1 - \phi q_{b}}}_{\text{liquidity services}}\right) m_{b}.$$
 (20)

We assume that bubbles grow at the same pace as the economy as long as they persist.¹²

¹²Focusing on the stationary bubble is a standard practice in the literature. As discussed in previous studies (Tirole, 1985; Farhi and Tirole, 2012; Hirano and Yanagawa, 2017), there are multiple equilibria with bubbles in this class of models. Specifically, there is a continuum of asymptotically bubbleless equilibria, in each of which the bubble starts at a smaller size than the stationary one, then shrinking relative to the economy and converging to zero even if the economy is staying in the bubbly regime. The stationary bubble is the largest bubble that can be sustained in equilibrium; if the bubble starts at a larger size, it grows faster than the economy, and eventually becomes too large to be sustained.

We solve equations (18), (19), and (20) for endogenous variables,¹³ finding

$$m_b = \frac{r\beta}{\beta\pi + 1 - \beta + \sigma_b\beta\left(\phi - \pi\right)} \left[2 - \pi - \frac{1}{\beta} - \left(\phi + \sigma_b\left(1 - \pi\right)\right)\right].$$

From this equation, it is clear that for positive bubbles to exist, $\phi + \sigma_b (1 - \pi) < 2 - \pi - \frac{1}{\beta}$ has to be satisfied. Let us discuss this condition intuitively. First, the financial constraint has to be tight (low ϕ); otherwise, liquidity services are low, and bubbles have to offer high capital gain to be held, eventually becoming too large to be sustained. Second, bubbles cannot be too risky; otherwise, they have to compensate for the risk by capital gain, eventually becoming too large to be sustained. The bursting probability σ_b has to be smaller than $\bar{\sigma}_b \equiv \frac{1}{1-\pi} \left[2 - \pi - \frac{1}{\beta} - \phi\right]$ for m_b to be positive. We assume that this condition is satisfied throughout this section.

We can rewrite the existence condition using the structure of the economy. Namely, m_b is strictly positive if and only if

$$(1 - \sigma_b) \underbrace{g_f^*}_{\text{growth rate}} > \underbrace{\frac{r}{q_f^*}}_{\text{interest rate}}$$

If bubbles are deterministic ($\sigma_b = 0$), there can be a bubbly equilibrium if and only if the growth rate is greater than the interest rate in the bubbleless economy. The condition is harder to be satisfied if bubbles are risky, because risky bubbles have to offer high capital gain to exist and hence are easier to explode.

Combining equations (16) and (19), we find that capital growth during the bubbly episode is given by

$$\underbrace{g_b}_{\text{bubbly growth}} = \left(1 + \underbrace{\frac{m_b}{r}}_{\text{crowding-in effect}}\right) \left(1 + \underbrace{\frac{-\phi}{1 - \phi q_b} \left(q_f^* - q_b\right)}_{\text{crowding-out effect}}\right) \underbrace{g_f^*}_{\text{growth after bursting}}.$$

This equation shows that realized bubbles have two effects on growth. First, realized bubbles crowd investment in because they provide liquidity to investors. This effect is strong if the financial constraint is tight; the proof is in Section D.3 in the appendix. Second, realized bubbles crowd demand away from capital and reduce its price; the poof is in Section D.4 in the appendix. Low capital price is harmful to investment because it decreases investors' wealth and in addition increases $1-\phi q$, which is the "down payment" investors have to pay to conduct a unit of investment. The crowding-out effect, however, is weak if ϕ is small, because investors do not rely on capital to

¹³We assumed that $m_b \neq 0$ and obtained the solution. $\{m_b, q_b, g_b\} = \{0, q_f^*, g_f^*\}$ solve equations (18), (19), and (20) too. The bubbleless equilibrium exists because bubbly assets are intrinsically useless. Needless to say, we are interested in a bubbly equilibrium now.

obtain liquidity much. Hence, bubbles enhance growth if the financial constraint is tight, because the crowding-in effect is large, while the crowding-out effect is moderate. We summarize the result in the following proposition. The proof is in Section D.5 in the appendix.

Proposition 1 In the simple model described above, suppose that the economy is now in the bubbly regime; i.e., the economy has a stochastic bubble. Then, there exists a value of ϕ below which the crowding-in effect dominates the crowding-out effect. Asset bubbles are growth-enhancing in this parameter region. If ϕ is larger than the threshold value, the crowding-out effect dominates and asset bubbles are growth-reducing.

Finally, we consider the fundamental regime before the bubbly episode starts. The feasibility constraint for investment is written as

$$(1 - \phi q_f) g_f = \pi r. \tag{21}$$

Combining equations (16) and (21), we find that the growth rate of capital in the fundamental regime before the bubbly episode is written as

$$\underbrace{g_f}_{\text{growth before emergence}} = \left(1 + \underbrace{\frac{-\phi}{1 - \phi q_f} \left(q_f^* - q_f\right)}_{\text{crowding-out effect}}\right) \underbrace{g_f^*}_{\text{growth after bursting}}.$$

There is the crowding-out effect alone. We obtain the following proposition.

Proposition 2 In the simple model described above, suppose that the economy is now in the fundamental regime before the bubbly episode. Bubbles are expected to arise at a positive probability. Then, the growth rate of capital is lower than the one in the fundamental regime after the bubbly episode in which no future bubbles are expected. Moreover, the growth rate of capital before the bubbly episode is reduced if the bubbly episode arises at a higher probability.

Proof. In the fundamental regime before the bubbly episode, the Euler equation for capital is written as

$$q_f = \frac{r}{g_f} \left[\left(1 - \sigma_f\right) \beta \left(1 + \pi \frac{q_f - 1}{1 - \phi q_f}\right) + \sigma_f \beta \left(\frac{\hat{c}_f}{\hat{c}_b}\right) \left(1 + \pi \frac{q_b - 1}{1 - \phi q_b}\right) \right],\tag{22}$$

where \hat{c}_f satisfies $\hat{c}_f = r - g_f$. We solve equations (21) and (22) for q_f and g_f , obtaining

$$q_f = \frac{(1-\pi)\beta + \sigma_f (1-\pi) \left(-\beta + \frac{1}{1-\sigma_b}\frac{r}{\hat{c}_b}\right)}{\pi (1-\beta) + \beta\phi + \sigma_f \left(\beta (\pi-\phi) + \frac{\phi}{1-\sigma_b}\frac{r}{\hat{c}_b}\right)}.$$
(23)

Taking a derivative, we obtain

$$\frac{\partial q_f}{\partial \sigma_f} = \left(\frac{1-\pi}{\pi \left(1-\beta\right) + \beta \phi + \sigma_f \left(\beta \left(\pi-\phi\right) + \frac{\phi}{1-\sigma_b} \frac{r}{\hat{c}_b}\right)}\right)^2 \frac{\beta \pi}{\left(1-\sigma_b\right) \left(1-\phi\right)} \left(\sigma_b - \bar{\sigma}_b\right),$$

which is negative because we assume that σ_b is smaller than $\bar{\sigma}_b$. Hence, q_f decreases with σ_f , and so does $g_f = \frac{\pi r}{1 - \phi q_f}$. In addition, as is clear from equations (17) and (23), q_f converges to q_f^* as σ_f converges to zero. Because we assume that σ_f is positive, q_f is smaller than q_f^* , and so is $g_f = \frac{\pi r}{1 - \phi q_f}$ than $g_f^* = \frac{\pi r}{1 - \phi q_f^*}$.

The top panel of Figure 1 illustrates the growth rate of capital before, during, and after the bubbly episode, displaying time along the horizontal axis. The crowding-in effect dominates the crowding-out effect of the realized bubble in this example, raising capital growth during the bubbly episode. Importantly, capital growth before the bubbly episode is lower than the one after the bubbly episode, and this result is robust to the parameter values. Intuitively, it is caused by the wealth effect of the future bubble. An asset bubble will increase wealth once it emerges in the future. Expecting its emergence, people increase consumption even before the bubbly episode begins, which crowds demand away from capital and reduces its price, thereby reducing the speed of capital accumulation. A bubble that has not been realized yet will have an impact on the current economy through expectations.

3.2 Stock Market to GDP Ratio

We discuss the stock market to GDP ratio next, for this is an important variable for our empirical investigation. The stock market value is defined as follows;

$$stock_t \equiv \phi \left[q_t K_{t+1} \right] + \tilde{p}_t \mathbf{1}_{\{z_t=b\}} M.$$

In Section B in the appendix, we discuss the micro-foundations of the financial frictions following Kiyotaki and Moore (2012). The microstructure allows us to define the stock market value clearly. Not all of the capital stock is publicly traded in our model,¹⁴ but the fraction of capital traded in the stock market increases with financial development measured by ϕ . This implication is consistent with the empirical literature (Sahay et al., 2015). In contrast, asset bubbles are traded in the stock market, because they are attached to the equity issued by final goods firms. This is an interpretation of bubbly assets proposed by Tirole (1985).¹⁵ Specifically, we assume that a new final goods firm is established when the economy switches to the bubbly regime, and it issues M units of equity and distributes them to the households exogenously. The firm's equity

¹⁴This is due to the lack of the investors' commitment power. Please see Section B in the appendix for details.

¹⁵He wrote: "In reality some of these "useless" pieces of paper are likely to pertain to [...] that of some firms producing with a constant returns to scale technology that is freely available" (see Tirole, 1985, pp. 1502).



Figure 1: Dynamics in Simple Model

is intrinsically useless because it makes zero profits from production, but nonetheless serves as a bubbly asset traded in the stock market.

The bottom panel of Figure 1 illustrates the stock market to GDP ratio before, during, and after the bubbly episode. It rises when the bubble emerges, and collapses when the bubble bursts. This result is robust to the parameter values. We analytically prove it in Section D.7 in the appendix, and summarize it as a proposition.

Proposition 3 In the simple model described above, the stock market to GDP ratio rises when the bubble emerges, and collapses when the bubble bursts.

3.3 Comovement Problem in the Simple Model

While the simple model's tractability is useful to obtain analytical solutions, it has an empirically relevant limitation: it is unable to generate a boom when the bubble emerges, or a recession when the bubble bursts as we prove in Section D.8 in the appendix. The intuition is simple. Because we set $\alpha = 1$, Y_t/K_t is constant. Because output is predetermined, an increase in investment must be offset by a decrease in consumption to satisfy the goods market clearing condition. In short, the simple model suffers from a comovement problem.

To overcome this problem, we need mechanisms to change output in the short run.¹⁶ Our

¹⁶Hirano and Yanagawa (2017) change productivity in the short run. There are both productive and unproductive investors in their model. Hence, shifting resources between the two groups acts like a productivity shock. A similar mechanism is seen in Ajello (2014).

Parameter	Value	Calibration Target	
β	0.99	Exogenously Chosen	
lpha	0.33	Capital Share=0.33	
ζ	0.33	Comin and Gertler (2006)	
π	0.06	Shi (2015)	
δ_0	0.001	Frictionless Growth $g^4 = 1.02$	
$\delta_1 u^{1+\zeta}$	0.065	Frictionless Depreciation $\delta(u) = 0.05$	
η	2.67	Frictionless Hours $l = 0.27$	
$\bar{A}u^{lpha}$	0.49	Equilibrium Condition	
u	1	Normalization	

Table 1: Parameters and Calibration Targets

baseline model has both an endogenous labor supply and a variable capacity utilization rate for this purpose, both of which are standard in the business cycle literature. They play crucial roles in overcoming the comovement problem as we explain in Section 5.

4 Calibration

Table 1 summarizes the parameter values used for the quantitative analysis. We set the discount factor at $\beta = 0.99$, the capital share at $\alpha = 0.33$, and the elasticity of $\delta'(u_t)$ at $\zeta = 0.33$, following Comin and Gertler (2006). The probability of having an investment opportunity is set at $\pi = 0.06$, following Shi (2015).

The rest of the parameters are calibrated in the model. We assume that if the financial constraint is sufficiently loose and does not bind, the growth rate of the economy would be 2% per year, hours worked would be 27% of the available time, and the depreciation rate would be 5% per quarter along the balanced growth path. We then solve for the three parameters δ_0 , $\delta_1 u^{1+\zeta}$, and η jointly. We find the value of $\bar{A}u^{\alpha}$ from the equilibrium condition. We set u = 1, which is just a normalization.

One may find that the target depreciation rate (5% per quarter) is high, but remember that this is the depreciation rate in an extreme situation in which the financial constraint never binds. Previous studies in the literature assume that the financial constraint is relevant. If we follow Kiyotaki and Moore (2012) and set it at $\phi = 0.19$, the implied depreciation rate in the bubbleless equilibrium is 2.4% per quarter. However, we are agnostic about the value of ϕ at this point. We show comparative statics with respect to this parameter in the following section.

5 Comparative Statics

In this section, we discuss the implications of recurrent bubbles in the calibrated model. To focus on their role, we shut down the supply and demand shocks $a_t = d_t = 0$ for all $t \ge 0$ throughout this section.

5.1 Growth Implications

The blue line in Figure 2 shows the growth rate of capital if the economy is always in the fundamental regime. Relaxing the financial constraint (horizontal axis) raises the growth rate of capital if it is initially tight. But interestingly, relaxing the financial constraint decreases the growth rate of capital if it is initially loose. For this result, the endogenous capacity utilization rate is important. Capital is relatively cheap in an economy with an advanced financial system as shown in Figure 3. Low capital price induces households to choose a high utilization rate because the opportunity cost to do so is low, reducing both net investment and capital growth.

The green diamonds in Figure 2 show the growth rate of capital if the economy has a stochastic bubble. We assume that the probability of bursting is 1.5% per quarter, but the results are robust to other choices. The stochastic bubble raises the growth rate of capital except for a small parameter region near the existence threshold. This result is consistent with Proposition 1 in Section 3.

Finally, the red circles and crosses in Figure 2 show the regime-dependent growth rate of capital in the recurrent-bubble equilibrium. We assume that the probabilities of regime switches are 1.5% per quarter in both directions, but the results are robust to other choices. We see both the crowding-in effect of realized bubbles and the crowding-out effect of future bubbles. Because of the crowding-in effect of realized bubbles, capital accumulation is faster in the bubbly regime than in the fundamental regime. In addition, because of the crowding-out effect of future bubbles, capital accumulation is slower in the recurrent-bubble equilibrium than in the stochastic-bubble equilibrium, conditional on being in the same regime (red circle versus green diamond for the bubbly regime, and red cross versus green asterisk for the fundamental regime).

Figure 3 shows the same effects in detail. In the recurrent-bubble equilibrium, people invest more, consume more, work harder, and use capital more intensively in the bubbly regime than in the fundamental regime. Intuitively, the bubbly regime is a favorable time for investment, and households, recognizing it, optimally allocate both time and resources not only across time but also across regimes. This is the crowding-in effect of realized bubbles. In addition, people consume more, work less (spend more time on leisure), and invest less in the recurrent-bubble equilibrium than in the stochastic-bubble equilibrium, conditional on being in the same regime. People understand that future bubbles will make them wealthier, and this optimistic expectation makes them lazy now, loosely speaking. This is the crowding-out effect of future bubbles.¹⁷

¹⁷In reality, the wealth created by bubbles is unequally distributed in the economy. For example, a stock market boom will benefit equity holders disproportionately. Based on this observation, one may argue that the crowdingout effect of future bubbles is relevant to a small group of people and hence its impact on the macroeconomy is limited. We have two counterarguments to this claim. First, even though ex post direct beneficiaries are limited, ex ante optimism may be widespread, because it is hard to tell in advance exactly in which markets, in which assets,



Figure 2: Recurrent Bubbles and Economic Growth

Because the two effects offset each other, it is not obvious if recurrent bubbles raise long-run growth. The red triangles in Figure 2 show that in economies with under-developed financial systems, recurrent bubbles are beneficial to economic growth in long run.¹⁸ But in economies with advanced financial systems, recurrent bubbles are harmful to long-run growth. In both cases, the growth in the recurrent-bubble equilibrium is bumpy, because it is disrupted by the occasional bursting and emergence of bubbles.

5.2 Welfare Implications

The welfare impact of recurrent bubbles can be different from their growth impact for at least two reasons. First, there are other variables affecting utility, specifically, leisure and consumption. Second, volatility matters for utility too. This section takes these factors into account and evaluates the welfare impact of recurrent bubbles.

$$\bar{g} = g_f^{\frac{\sigma_b}{\sigma_f + \sigma_b}} g_b^{\frac{\sigma_f}{\sigma_f + \sigma_b}}$$

where g_b and g_f denote capital growth in the bubbly and fundamental regimes, respectively.

and in what forms bubbles will appear and develop. Second, the potential beneficiaries may be more productive and wealthier on average, and if so, their decisions will have large macroeconomic impacts even though they are relatively small in population.

 $^{^{18}}$ Long-run growth is given by



Figure 3: Recurrent Bubbles and Macroeconomic Variables

Figure 4 plots the welfare levels derived from the utility function.¹⁹ It resembles Figure 2, suggesting the importance of economic growth as a determinant of welfare. Recurrent bubbles improve welfare if financial systems are under-developed, but the opposite is true if financial systems are developed. Interestingly, bubbly-regime welfare in the stochastic-bubble equilibrium (green diamonds) is higher than bubbly-regime welfare in the recurrent-bubble equilibrium (red circles) even though people in the former equilibrium will have fewer bubbly episodes in the future (in fact, none). This observation highlights that the stochastic bubble is special; the bubble exists from the beginning, but the analysis abstracts from the periods before the bubbly episode starts. Therefore, it inevitably emphasizes the crowding-in effect of realized bubbles. This is part of the reason why bubbles tend to be welfare-improving in the literature. In contrast, the analysis with the recurrent bubbles considers periods before bubbly episodes start too. As a result, it gives us a more balanced view of bubbles, for the crowding-out effect of future bubbles naturally emerges and offsets the crowding-in effect.

The tradeoff between the crowding-in effect of realized bubbles and the crowding-out effect of future bubbles becomes even more transparent by analyzing the welfare impact of high-frequency bubbles.²⁰ Specifically, we change the frequency of bubbles (σ_f) while keeping the other parameters, including the probability of bursting, constant. Results are shown in Figure 5. Blue signs show the welfare gains of high-frequency bubbles, and pink signs show the welfare gains of high-frequency bubbles, ²¹

We see a shape like a flounder. The fact that its belly is blue and its back is pink implies that the welfare-maximizing frequency of bubbles decreases with the level of financial development. The intuition is simply put. If the economy's financial system is severely under-developed, highfrequency bubbles are preferred because they can mitigate the liquidity shortage. But as the

 19 We rewrite the utility function (8) in the recursive form and detrend it, obtaining

$$\hat{V}_t = (1 - \beta) \left\{ \log \left[\hat{c}_t \right] + (1 - \pi) \eta \log \left[1 - l_t \right] \right\} + \beta \log \left[g_t \right] + \beta E_t \left[\hat{V}_{t+1} \right].$$

 \hat{V}_t is defined as $\hat{V}_t \equiv V_t - \log K_t$, where V_t is the continuation utility value. We calculate the regime-dependent welfare levels by solving the following equations:

$$\begin{pmatrix} \hat{V}_f \\ \hat{V}_b \end{pmatrix} = \begin{pmatrix} (1-\beta) \left\{ \log \left[\hat{c}_f \right] + (1-\pi) \eta \log \left[1 - l_f \right] \right\} + \beta \log \left[g_f \right] \\ (1-\beta) \left\{ \log \left[\hat{c}_b \right] + (1-\pi) \eta \log \left[1 - l_b \right] \right\} + \beta \log \left[g_b \right] \end{pmatrix} + \begin{pmatrix} 1-\sigma_f & \sigma_f \\ \sigma_b & 1-\sigma_b \end{pmatrix} \begin{pmatrix} \beta \hat{V}_f \\ \beta \hat{V}_b \end{pmatrix},$$

where the subscripts f and b denote the fundamental and bubbly regimes, respectively. We calculate the unconditional welfare level in the recurrent-bubble equilibrium by

$$\bar{\hat{V}} \equiv \frac{\sigma_b}{\sigma_b + \sigma_f} \hat{V}_f + \frac{\sigma_f}{\sigma_b + \sigma_f} \hat{V}_b.$$

We plot \hat{V}_f , \hat{V}_b , and $\bar{\hat{V}}$ in Figure 4. Without loss of generality, we subtract the welfare level in an economy in which the financial constraint is too loose to bind when we plot them.

 20 The authors thank Jean Tirole for the discussion that led us to this exercise.

²¹We plot $\bar{\hat{V}}(\sigma_f) - \bar{\hat{V}}(1.5\%)$ in Figure 5, where $\bar{\hat{V}}(\sigma_f)$ is the unconditional welfare level in the recurrent-bubble equilibrium given σ_f .



Figure 4: Recurrent Bubbles and Welfare

financial system gradually develops, the liquidity shortage becomes less important, and instead, the crowding-out effect of future bubbles emerges as a new problem. Low-frequency bubbles are preferred in this situation, because the crowding-out effect of future bubbles is weak if bubbles emerge less frequently (see Proposition 2 in Section 3).

5.3 Comovement

Remember that the simple model discussed in Section 3 had the comovement problem. This section shows that our baseline model is able to overcome it. Analytically, output growth in the recurrent-bubble equilibrium is given by

$$\frac{Y_t}{Y_{t-1}} = \frac{\hat{Y}_t}{\hat{Y}_{t-1}} \frac{K_t}{K_{t-1}} = \frac{\hat{Y}_t}{\hat{Y}_{t-1}} g_{t-1} = \begin{cases} g_f, & \text{if } \{z_{t-1}, z_t\} = \{f, f\}, \\ \frac{\hat{Y}_b}{\hat{Y}_f} g_f, & \text{if } \{z_{t-1}, z_t\} = \{f, b\}, \\ g_b, & \text{if } \{z_{t-1}, z_t\} = \{b, b\}, \\ \frac{\hat{Y}_f}{\hat{Y}_b} g_b, & \text{if } \{z_{t-1}, z_t\} = \{b, f\}, \end{cases}$$

where \hat{Y}_t is output to capital ratio ($\hat{Y}_t \equiv Y_t/K_t$), and subscripts *b* and *f* indicate that the variables are determined in the bubbly and fundamental regimes respectively. As shown in Figure 3, \hat{Y}_b is larger than \hat{Y}_f , because both labor supply and the capacity utilization rate are higher in the



Figure 5: Bubble Frequency and Welfare

bubbly regime than in the fundamental regime. As a result, output growth rises when bubbles emerge (in period t with $\{z_{t-1}, z_t\} = \{f, b\}$), and plunges when bubbles burst (in period t with $\{z_{t-1}, z_t\} = \{b, f\}$). Similar arguments hold for both consumption and investment. Therefore, the economy has a recession when a bubble bursts, and a boom when a bubble emerges.

The stock market value moves in the same way. As shown in Figure 3, the stock market value to capital, $stock_t/K_t$, is larger in the bubbly regime than in the fundamental regime. Applying the same argument as we applied to output growth, we see that the stock market value drops when a bubble bursts, and rises when a bubble emerges. This is an important observation for our empirical investigation, which we discuss next.

6 Taking the Model to the Data

In this section, we revisit the recent U.S. economic data through the lens of our model. We are particularly interested in identifying when the economy had a bubble and when it did not.

6.1 Data

We use quarterly U.S. data on GDP growth and the stock-market-to-GDP ratio for the period 1984.Q1-2017.Q4 to estimate the likelihood of bubbles as well as the paths of productivity and preference shocks in our model, which is discussed at length in Section G in the appendix. We

choose these observables because our model has robust predictions about them. That is, not only is GDP growth high but also the stock market booms when bubbles exist as we discussed in the previous sections.

Figure 6 shows the observables. It is clear from the 10-year rolling-window average (red line) that GDP growth is slowly declining, going from 0.7% (2.8% in annual terms) in the 1990s, to 0.87% (3.5%) in 2005, to less than 0.4% (1.6%) after the Great Recession. Three boom episodes in the stock market-to-GDP ratio are also clear: before Black Monday in 1987, the IT bubble in the late 1990s, and before the Great Recession.²² Our identification strategy exploits the connection between these two variables to uncover the presence of bubbles in the data.

For example, during the years leading up to the Great Recession, GDP growth was high, averaging 3% per year (black circles), and the stock-market-to-GDP ratio expanded aggressively. We observe the opposite during the post-crisis years: lackluster growth of 1.6% per year (black diamonds) and a sharp contraction in the stock market. These observations suggest that the economy was in the bubbly regime before 2007, but the bubble crashed in 2008, and since then, the economy has not had a new bubble. But this is only one possibility. It is also possible that the ups and downs in GDP growth and the stock market had nothing to do with bubbles but were driven by real factors, namely, productivity and preference shocks. The estimation exercise is informative, for it tells us the likeliest account in light of the data.

6.2 Method

The model is estimated using Bayesian methods (Fernandez-Villaverde et al., 2016) and a nonlinear filter (Kim and Nelson, 1999). We assume that the economy is in the bubbly equilibrium (see Section H in the appendix for details on the solution and estimation of the model).²³ We impose the condition that the productivity and preference shocks follow an AR(1) process, and estimate the persistence, ρ_i , and standard deviation, SD_i , of these stochastic processes ($i \in \{\text{productivity} (a), \text{ preference } (d)\}$). Except for ϕ , σ_f , and σ_b , all parameter values are those calibrated and reported in Table 1. We set both σ_f and σ_b at 1.5% as in the previous section. Recall that we treated ϕ as a free parameter in the previous section to inspect the model's mechanisms. In this section, we choose $\phi = 0.19$ as in Kiyotaki and Moore (2012).

We also estimate the regime-dependent average capital growth. Specifically, we assume that it is determined by the sum of the model's implied capital growth $(\mu_{g,z}^m)$ and a constant $(\bar{\mu}_{g,z})$, and estimate $\mu_{g,z} = \mu_{g,z}^m + \bar{\mu}_{g,z}$ for $z \in \{f, b\}$. The reason is the following. Our model predicts that capital growth is high in the bubbly regime. Once calibrated, it has a quantitative prediction too,

 $^{^{22}}$ Stock-market-to-GDP ratio is de-trended. The resulting data indicates that this ratio was 20% above its HP-filtered trend before the Great Recession. By 2009, the ratio was almost 35% below its trend.

 $^{^{23}}$ Our model falls within the class of MS-DSGE models discussed in Farmer et al. (2009). We find a minimumstate-variable equilibrium. The absence of endogenous state variables greatly simplifies the solution method, as otherwise we would have to rely on the methods in Farmer et al. (2011).



Figure 6: GDP Growth and Stock Market-GDP Ratio in Data

but it may not match the "data counterpart," which would be calculated from the data only if we perfectly knew when bubbles existed in the U.S. economy. Moreover, we do not want to use them as calibration targets, because that exercise needs to take an a priori stance on the timing of the regime switch before estimating it. To work around this issue, we estimate the regimes and the regime-dependent average capital growth at the same time. A similar strategy is imposed on the regime-dependent average stock-market-to-GDP ratio.

6.3 Priors and Posteriors

Table 2 presents both the priors and posteriors (mode and 90% credible bands). We impose standard beta and inverse-gamma priors for parameters regarding the productivity and preference shocks (see Fernandez-Villaverde et al. (2015) for priors on persistence parameters and Fernandez-Villaverde et al. (2016) for priors on volatility parameters). We use normal priors for the means of capital growth and the stock-market-to-GDP ratio in the fundamental and bubbly regimes, $\{\mu_{g,f}, \mu_{stock/GDP,f}, \mu_{g,b}, \mu_{stock/GDP,b}\}$, respectively. The choice of their prior means is guided by the model's robust predictions that both capital growth and the stock-market-to-GDP ratio are high in the bubbly regime.

The priors and posteriors are different, which points to the informativeness of the data. Importantly, the posterior modes indicate that both capital growth and the stock-market-to-GDP ratio

Parameter	Prior	Posterior
$\mu_{g,f}$	N(0.5, 0.1)	$\underset{\left[0.56,0.75\right]}{0.65}$
$\mu_{stock/GDP,f}$	N(0.0, 1)	-1.67 [-2.95,-0.45]
$\mu_{g,b}$	N(0.75, 0.1)	$\underset{[0.63,0.93]}{0.78}$
$\mu_{stock/GDP,b}$	N(10, 1)	$\underset{[8.81,11.87]}{10.25}$
$ ho_d$	B(0.15, 0.05)	$\underset{\left[0.40,0.58\right]}{0.50}$
SD_d	IG(6,1)	$\underset{[0.07,0.09]}{0.08}$
$ ho_a$	B(0.75, 0.05)	$\underset{[0.85,0.92]}{0.89}$
SD_a	IG(6,1)	$\underset{\left[0.01,0.02\right]}{0.01}$

 Table 2: Estimated Parameters

are higher in the bubbly regime. For example, the average capital growth is estimated to be about 52 basis points higher in annual terms in the bubbly regime. In terms of the structural shocks, the preference disturbance is volatile but lacks persistence. The productivity shock is relatively persistent and stable.²⁴ Interestingly, the estimated persistence of the productivity shock (ρ_a) is lower than the typical number in the literature (≈ 0.95). This is because both the endogenous growth mechanism and the regime switching add persistence to the model.

6.4 Impulse Response Functions

Figure 7 plots impulse response functions to both productivity and preference shocks. Responses to one-standard-deviation shocks are plotted. The two shocks cause distinct dynamics on the observables. A positive productivity shock (a rise in a_t) raises GDP growth temporarily but has a mild impact on the stock-market-to-GDP ratio. This is because the shock raises both GDP and the stock market value simultaneously. In contrast, a positive preference shock (a rise in d_t) raises the stock market-to-GDP ratio by making people effectively patient, but its impact on GDP growth is weak in the short run.²⁵ Becoming patient, households increase investment, which raises GDP growth in subsequent periods. Impulse response functions are modestly regime-dependent.²⁶ In Section I in the appendix, we report the responses of other variables.

 $^{^{24}}$ We tried alternative means and standard deviations for the priors. Our results are robust to these variations. This is not surprising given how tightly estimated the parameters are.

 $^{^{25}}$ After the shock, households end up putting large weights to the utility flows in the distant future relative to those in the near future because the preference shock is mean reverting.

²⁶We do not consider the effect of regime switch in Figure 7. For example, the solid blue line is plotted under the assumption that $z_t = b$ for $1 \le t \le 8$.



Figure 7: Impulse Response Functions

6.5 Estimated Regimes and Shocks

Figure 8 presents the estimated probability of the bubbly regime. The economy spent most of the time in the fundamental regime before 1997, with a brief exception prior to 1987, which was abruptly terminated by Black Monday. Relatively strong economic growth during the first 10 years or so of the sample was mainly driven by the real shocks plotted in Figure 9. This result is not surprising given the moderate stock-market-to-GDP ratio observed in the data in this period. During the second half of the 1990s, a combination of positive productivity shocks and the emergence of the bubble raised both GDP growth and stock market value. This bubble started around 1997 and ended in the second quarter of 2001 according to our estimate. Because of this timing, we call it the "IT bubble."²⁷

After the IT bubble crash, the probability of the bubbly regime rose again in 2006, raising both GDP growth and the stock market. By mid-2006, the probability exceeded 50%, and between 2007 and mid-2008, it came close to 100%. We call it the "housing bubble" because of its timing. Importantly, robust growth in this period is mainly driven by the bubble; notice that productivity shocks are unfavorable in this period. This is different from the economic boom in the mid-1990s, which was driven by favorable real shocks. The Great Recession was caused by the collapse of the housing bubble and highly unfavorable real shocks. Particularly large and adverse preference

²⁷Strictly speaking, our one-sector model has no reason to connect this bubble to the information technology sector. We nonetheless use this term for convenience. The same applies to the "housing bubble."

shocks were observed in 2008. We think that they reflect adverse financial shocks that we do not model explicitly in this paper.²⁸

The return to the fundamental regime mixed with adverse productivity shocks explains the lackluster performance of the US economy during the last decade. In the final part of the sample, our approach assigns some probability that the economy experienced a new bubble. In the data, growth was relatively strong in 2014, and so was the stock market in the midst of ultra-loose monetary policies around the world. But as the Fed moved to normalize its monetary policy in 2015, both the stock market and GDP growth cooled down. Our model concludes that the evidence is not strong enough to judge that a bubble was present. The chance was less than 50% according to our estimate.

As we can see, the estimation exercise delivers bubbly regimes that are consistent with the historical narrative. But we can say more about the quantitative predictions of our theory. For instance, one implication of the model is that the labor market is strong during the bubbly regime because of the crowding-in effect (Figure 3). Consistent with this prediction, employment in the data had peaks of 1.7 pp and 2.5 pp above trend in the late 1990s and before the Great Recession, respectively. A similar validation is possible if one considers the dynamics of capacity utilization, which is above average during bubbles both in the theory and the data.

It seems natural to ask whether the estimated regimes coincide with other relevant economic shifts in the U.S. To answer this question, we compare our results to three classes of regimes reported in the literature. First, the estimated regimes in Figure 8 are different from those in Sims and Zha (2006), who fit US data to a regime-switching VAR with drifting coefficients and variances. They report the existence of four distinct regimes: the Greenspan state prevailing during the 1990s and early 2000s; the second most common regime emerges in the early 1960s and parts of the 1970s; the last two regimes correspond to sporadic events such as 9/11. Our estimates are not similar to those estimated to account for the Great Moderation, with a high-volatility regime prior to 1984 and a calmer one post-1984 (Stock and Watson, 2002). Finally, our estimates bear little resemblance to recession regimes (Chauvet and Piger, 2008). See Hamilton (2016) for an extensive review of regime switching in macroeconomics.

Before moving to the counterfactual exercises, we want to stress that our findings are robust to alternative calibration and identification strategies. They include 1) using the liquidity parameter ϕ to match the means of GDP growth and the credit-to-GDP ratio in and out of the Great Recession, with the caveat that we impose the dates when bubbles exist a priori; 2) a longer sample; 3) using credit market data or the Shiller-Case house price index in lieu of the stock market data; 4) a third regime featuring high growth and a high credit-to-GDP ratio driven by non-bubble forces; and 5) GDP growth and the consumption-to-investment ratio as observables. See Section J in the appendix for details.

²⁸See Guerron-Quintana and Jinnai (2019) for more discussion on the financial shock and the Great Recession.



Figure 8: Estimated Probability of Bubbly Regime



Figure 9: Estimated Preference and Productivity Shocks

6.6 Counterfactual Simulations

How important were bubbles for the U.S. economy? We answer this question by two counterfactual simulations. The first one is the "no-bubble-by-chance" simulation, in which the probability of the economy being in the bubbly regime was artificially set at zero. But the economy is still in the recurrent-bubble equilibrium; it is only the realization of the regime that we change. The dashed red line in Figure 10 shows the trend of log GDP under this counterfactual scenario.²⁹ The solid blue line shows the GDP trend in the benchmark scenario. The bubble has both short-run and middle-run impacts on the economy. The short-run impact is the economic boom directly caused by the realized bubble, i.e., a "plateau" made by the solid blue line and the dashed red line in the middle of each panel. According to our estimate, the short-run impact is already sizable; by about 6 to 7 percentage points more goods and services were produced during the bubbly booms.

The medium-run impact is visually subtle but economically important. Notice that the solid blue line is higher than the dashed red line even after the bubble is gone. This is because the solid blue line has a steeper slope than the dashed red line during bubbly episodes, which is a graphical confirmation that capital growth is higher in the bubbly regime than in the fundamental regime. As for the IT bubble, the GDP trend is about 1.2 percentage points higher in the baseline scenario in the years after the bubble burst. For the housing bubble, it is about 60 basis points. Combined, the two bubbles permanently raised the level of U.S. GDP by about 2 percentage points more than in the scenario in which bubbles were not realized by chance.

Our second exercise is the "no-chance-of-bubble" simulation, by which we mean that the economy is in the fundamental equilibrium. Hence, bubbles were neither realized nor expected to do so. The trend line under this scenario corresponds to the dotted black line in Figure 10. Clearly, the economy would have grown at the fastest pace in this scenario. This is because of the absence of the crowding-out effect of future bubbles; had people not expected bubbles to emerge, they would have consumed less, worked more, and invested more, all of which would have contributed to higher growth. So our model suggests that realized bubbles are better than no realization, but if we could move to a different equilibrium in which economic agents do not expect bubbles, that would be better.³⁰

7 Conclusions

We have developed a model with recurrent bubbles, crashes, and endogenous growth in an infinite horizon economy. Theoretically, we have shown that there is a novel crowding-out effect of future

²⁹It is normalized to 0 in 1997 in the upper panel and in 2005 in the lower panel. To facilitate the comparison across simulations, we shut down the structural shocks. Its impact on the result is minor because responses to structural shocks are only mildly regime-dependent (see Figure 7).

³⁰The economy in the fundamental equilibrium would have grown faster in the long run even if either the IT or the housing bubble had never burst. This is possible because in the recurrent-bubble equilibrium, the crowding-out effects of both the realized bubble and future bubbles offset the crowding-in effect of the realized bubble.



Figure 10: Counterfactual Simulations

bubbles because infinitely lived households anticipate that bubbles will emerge in the future. Empirically, we estimate the model using U.S. data for the period 1984-2017 and find that at least two bubbly episodes are very likely in the sample. The U.S. economy benefited from these bubbles, but our model suggests that it would have grown faster in the long run if it had been in a different equilibrium in which bubbles never arose and were never expected to emerge.

One direction for our future research would be examining government policies. Various forms of "leaning-against-the-bubble" policies are particularly interesting to study, which both economists and policymakers have been discussing for many years (Barlevy, 2018). It would be fruitful to examine how the effects of such policies depend on the nature of recurrent bubbles such as frequency and duration. Because our model can be estimated, we can quantitatively assess the costs and benefits of the policies based on the estimated structural model too.

References

- AJELLO, A. (2014): "Financial Intermediation, Investment Dynamics, and Business Cycle Fluctuations," American Economic Review, 106, 2256–2303.
- ALLEN, F., G. BARLEVY, AND D. GALE (2021): "Asset Price Booms and Macroeconomic Policy: A Risk-Shifting Approach," *Forthcoming in American Economic Journal: Macroeconomics*.

- ANZOATEGUI, D., D. COMIN, M. GERTLER, AND J. MARTINEZ (2019): "Endogenous Technology Adoption and R&D as Sources of Business Cycle Persistence," *American Economic Journal: Macroeconomics*, 11, 67–110.
- ARROW, K. J. (1962): "The Economic Implications of Learning by Doing," The Review of Economic Studies, 29, 155–173.
- BARLEVY, G. (2018): "Bridging Between Policymakers' and Economists' Views on Bubbles," *Economic Perspectives*, 42, 1–21.
- BASSETT, W. F., A. DAIGLE, R. M. EDGE, AND G. I. KARA (2015): "Credit-to-GDP Trends and Gaps by Lender- and Credit-Type," Tech. Rep. December 3, 2015, FEDS Notes.
- BEAUDRY, P. AND F. PORTIER (2006): "Stock Prices, News, and Economic Fluctuations," *The American Economic Review*, 96, 1293–1307.
- BEWLEY, T. (1980): "The optimum quantity of money," in *Models of Monetary Economies*, ed. by J. Kareken and N. Wallace, 169–210.
- BISWAS, S., A. HANSON, AND T. PHAN (2020): "Bubbly Recessions," American Economic Journal: Macroeconomics, 12, 33–70.
- BLANCHARD, O., E. CERUTTI, AND L. SUMMERS (2015): "Inflation and Activity Two Explorations and their Monetary Policy Implications," Tech. Rep. 21726, NBER Working Papers.
- BRUNNERMEIER, M. K. AND M. OEHMKE (2013): "Bubbles, Financial Crises, and Systemic Risk," in *Handbook of the Economics of Finance*, ed. by G. M. Constantinides, M. Harris, and R. M. Stulz, Elsevier, vol. 2, 1221–1288.
- BRUNNERMEIER, M. K. AND Y. SANNIKOV (2014): "A Macroeconomic Model with a Financial Sector," *American Economic Review*, 104, 379–421.
- CERRA, V. AND S. C. SAXENA (2008): "Growth Dynamics: The Myth of Economic Recovery," American Economic Review, 98, 439–457.
- CHAUVET, M. AND J. PIGER (2008): "A Comparison of the Real-Time Performance of Business Cycle Dating Methods," *Journal of Business & Economic Statistics*, 26, 49–42.
- COMIN, D. AND M. GERTLER (2006): "Medium-Term Business Cycles," American Economic Review, 96, 523–551.
- DEL NEGRO, M., G. EGGERTSSON, A. FERRERO, AND N. KIYOTAKI (2017): "The Great Escape? A Quantitative Evaluation of the Fed's Liquidity Facilities," *American Economic Review*, 107, 824–857.

- DIBA, B. T. AND H. I. GROSSMAN (1988): "The Theory of Rational Bubbles in Stock Prices," *The Economic Journal*, 98, 754–746.
- FARHI, E. AND J. TIROLE (2012): "Bubbly Liquidity," *The Review of Economic Studies*, 79, 678.
- FARMER, R. E. A., D. F. WAGGONER, AND T. ZHA (2009): "Understanding Markov-switching rational expectations models," *Journal of Economic Theory*, 144, 1849–1867.
- (2011): "Minimal state variable solutions to Markov-switching rational expectations models," *Journal of Economic Dynamics and Control*, 35, 2150–2166.
- FERNANDEZ-VILLAVERDE, J., P. GUERRON-QUINTANA, AND J. F. RUBIO-RAMIREZ (2015): "Estimating dynamic equilibrium models with stochastic volatility," *Journal of Econometrics*, 185, 216–229.
- FERNANDEZ-VILLAVERDE, J., J. F. RUBIO-RAMIREZ, AND F. SCHORFHEIDE (2016): "Solution and Estimation Methods for DSGE Models," in *Handbook of Macroeconomics*, ed. by J. B. Taylor and H. Uhlig, Elsevier, vol. 2, 527–724.
- FISHER, J. D. M. (2014): "On the Structural Interpretation of the Smets-Wouters "Risk Premium" Shock," Tech. rep., Federal Reserve Bank of Chicago.
- GALI, J. (2014): "Monetary Policy and Rational Asset Price Bubbles," American Economic Review, 104, 721–752.
- (2016): "Insider-outsider labor markets, hysteresis and monetary policy," Tech. Rep. 1506, Department of Economics and Business, Universitat Pompeu Fabra.
- GERTLER, M. AND N. KIYOTAKI (2015): "Banking, Liquidity, and Bank Runs in an Infinite Horizon Economy," *American Economic Review*, 105, 2011–2043.
- GERTLER, M., N. KIYOTAKI, AND A. PRESTIPINO (2020): "A Macroeconomic Model with Financial Panics," *The Review of Economic Studies*, 87, 240–288.
- GREENWOOD, J., Z. HERCOWITZ, AND G. W. HUFFMAN (1988): "Investment, Capacity Utilization, and the Real Business Cycle," *American Economic Review*, 78, 417–402.
- GUERRON-QUINTANA, P. A. AND R. JINNAI (2019): "Financial frictions, trends, and the great recession," *Quantitative Economics*, 10, 735–773.
- HAMILTON, J. D. (2016): "Macroeconomic Regimes and Regime Shifts," in *Handbook of Macroeconomics*, ed. by J. B. Taylor and H. Uhlig, Elsevier, vol. 2, 163–201.

- HE, Z. AND A. KRISHNAMURTHY (2013): "Intermediary Asset Pricing," American Economic Review, 103, 732–770.
- HIRANO, T. AND N. YANAGAWA (2017): "Asset Bubbles, Endogenous Growth, and Financial Frictions," *The Review of Economic Studies*, 84, 406–443.
- IVASHINA, V. AND D. SCHARFSTEIN (2010): "Bank lending during the financial crisis of 2008," Journal of Financial Economics, 97, 319–338.
- JAIMOVICH, N. AND S. REBELO (2009): "Can News about the Future Drive the Business Cycle?" The American Economic Review, 99, 1097–1118.
- JORDA, O., M. SCHULARICK, AND A. M. TAYLOR (2015): "Leveraged bubbles," *Journal of Monetary Economics*, 76, S1–S20.
- KIM, C.-J. AND C. R. NELSON (1999): *State-Space Models with Regime Switching*, The MIT Press.
- KINDLEBERGER, C. P. (2001): Manias, Panics and Crashes, Wiley, 4th ed.
- KIYOTAKI, N. AND J. MOORE (2012): "Liquidity, Business Cycles, and Monetary Policy," Tech. Rep. 17934, NBER Working Papers.
- KOCHERLAKOTA, N. R. (1992): "Bubbles and constraints on debt accumulation," *Journal of Economic Theory*, 57, 245–256.
- (2009): "Bursting Bubbles: Consequences and Cures," Unpublished Manuscript, University of Minnesota.
- MARTIN, A. AND J. VENTURA (2012): "Economic Growth with Bubbles," *American Economic Review*, 102, 3033–3058.
- MENDOZA, E. G. (2010): "Sudden Stops, Financial Crises, and Leverage," *American Economic Review*, 100, 1941–1966.
- MIAN, A. AND A. SUFI (2015): House of Debt: How They (and You) Caused the Great Recession, and How We Can Prevent It from Happening Again, University of Chicago Press.
- MIAO, J., P. WANG, AND Z. XU (2015): "A Bayesian dynamic stochastic general equilibrium model of stock market bubbles and business cycles," *Quantitative Economics*, 6, 599–635.
- OKINA, K., M. SHIRAKAWA, AND S. SHIRATSUKA (2001): "The Asset Price Bubble and Monetary Policy: Japan's Experience in the Late 1980s and the Lessons: Background Paper," *Monetary and Economic Studies*, 19, 395–450.
- ROMER, P. M. (1986): "Increasing Returns and Long-Run Growth," Journal of Political Economy, 94, 1002–1037.
- SAHAY, R., M. CIHAK, P. M. N'DIAYE, A. BARAJAS, D. B. A. PENA, R. BI, Y. GAO, A. J. KYOBE, L. NGUYEN, C. SABOROWSKI, K. SVIRYDZENKA, AND S. R. YOUSEF (2015): "Rethinking Financial Deepening; Stability and Growth in Emerging Markets," Tech. Rep. 2015/008, IMF Staff Discussion Notes.
- SAMUELSON, P. A. (1958): "An Exact Consumption-Loan Model of Interest with or without the Social Contrivance of Money," *Journal of Political Economy*, 66, 467–482.
- SCHAAL, E. AND M. TASCHEREAU-DUMOUCHEL (2021): "Herding Through Booms and Busts," Unpublished Manuscript, Universitat Pompeu Fabra.
- SCHEINKMAN, J. A. AND L. WEISS (1986): "Borrowing Constraints and Aggregate Economic Activity," *Econometrica*, 54, 23–45.
- SCHMITT-GROHE, S. AND M. URIBE (2004): "Solving dynamic general equilibrium models using a second-order approximation to the policy function," *Journal of Economic Dynamics and Control*, 28, 755–775.
- ——— (2012): "What's News in Business Cycles," *Econometrica*, 80, 2764–2733.
- SHELL, K., M. SIDRAUSKI, AND J. E. STIGLITZ (1969): "Capital Gains, Income, and Saving," The Review of Economic Studies, 36, 26–15.
- SHESHINSKI, E. (1967): "Optimal accumulation with learning by doing," in *Essays on the Theory* of Optimal Economic Growth, ed. by K. Shell, MIT Press, 31–52.
- SHI, S. (2015): "Liquidity, assets and business cycles," *Journal of Monetary Economics*, 70, 116–132.
- SIMS, C. A. AND T. ZHA (2006): "Were There Regime Switches in U.S. Monetary Policy?" American Economic Review, 96, 54–81.
- SMETS, F. AND R. WOUTERS (2007): "Shocks and Frictions in US Business Cycles: A Bayesian DSGE Approach," American Economic Review, 97, 586–606.
- STOCK, J. AND M. WATSON (2002): "Has the Business Cycle Changed and Why?" NBER Macroeconomics Annual, 159–230.
- STOCK, J. H. AND M. W. WATSON (2012): "Disentangling the Channels of the 2007-09 Recession," *Brookings Papers on Economic Activity*, 81–141.

TIROLE, J. (1985): "Asset Bubbles and Overlapping Generations," Econometrica, 53, 1528–1499.

- TOWNSEND, R. M. (1980): "Models of Money with Spatially Separated Agents," in *Models of Monetary Economies*, ed. by J. Kareken and N. Wallace, 265–303.
- WEIL, P. (1987): "Confidence and the Real Value of Money in an Overlapping Generations Economy," *The Quarterly Journal of Economics*, 102, 1.
- WOODFORD, M. (1990): "Public Debt as Private Liquidity," *The American Economic Review*, 80, 382–388.

Appendices

NOT FOR PUBLICATION

A Household's Problem

A.1 With Loose Financial Constraints

This section solves the household's problem when financial constraints are loose. The financial constraint does not bind. We first show that the price of capital is equal to one in equilibrium. Suppose instead that it is greater than one. Then, the household can increase its utility by increasing i_t by Δ , and increasing both x_t^i and c_t^i by $(q_t - 1) \Delta$ for sufficiently small $\Delta > 0$. This is a contradiction to an equilibrium condition requiring that the household maximize its utility subject to the constraints.

Next, we show that the price of bubbly assets is equal to zero in the equilibrium. Suppose otherwise. Then, the Euler equation for bubbly assets,

$$\tilde{p}_t = E_t \left[\frac{\beta}{e^{d_{t+1}-d_t}} \left(\frac{c_t^i}{c_{t+1}^i} \right) \tilde{p}_{t+1} \mathbf{1}_{\{z_{t+1}=b\}} \right],$$

holds with a positive \tilde{p}_t in some t in the bubbly regime. To simplify the argument, we assume without loss of generality that $a_t = d_t = 0$ holds for all $t \ge 0$. Multiplying both sides by M and dividing them by K_t , we obtain

$$1 = (1 - \sigma_b) \beta \left(\frac{\hat{c}_{b,t}^i}{\hat{c}_{b,t+1}^i} \right) \left(\frac{m_{b,t+1}}{m_{b,t}} \right),$$

where $\hat{c}_{b,t}^i$ is the investor's consumption relative to the capital stock $(\hat{c}_t^i \equiv c_t^i/K_t)$ in the bubbly regime, and $m_{b,t}$ is the market value of the bubbly assets relative to the capital stock $(m_t \equiv \tilde{p}_t \mathbf{1}_{\{z_t=b\}}M/K_t)$ in the bubbly regime. To satisfy this equation, $(\frac{m_{b,t}}{\hat{c}_{b,t}^i})$ has to grow exponentially at the rate $(1 - \sigma_b)^{-1}\beta^{-1}$. But then, the transversality condition regarding bubbly assets is violated because

$$E_t \left[\beta^j \left(\frac{1}{c_{t+j}^i} \right) \tilde{p}_{t+j} \mathbf{1}_{\{z_{t+j}=b\}} M \right] = (1 - \sigma_b)^j \beta^j \left(\frac{m_{b,t+j}}{\hat{c}_{b,t+j}^i} \right)$$

does not converge to zero. This is a contradiction to an equilibrium condition requiring that the household maximize its utility subject to the constraints.

Because $q_t = 1$ and $\tilde{p}_t = 0$ hold if ϕ is large, the household's problem becomes standard. It

chooses a sequence of u_t , c_t^i , c_t^s , l_t , and n_{t+1} to maximize utility

$$E_0\left[\sum_{t=0}^{\infty} \frac{\beta^t}{e^{d_t}} \left(\pi \log\left(c_t^i\right) + (1-\pi)\left[\log\left(c_t^s\right) + \eta\left(1-l_t\right)\right]\right)\right]$$

subject to

$$\pi c_t^i + (1 - \pi) c_t^s + n_{t+1} = \left[u_t r_t + (1 - \delta (u_t)) \right] n_t + w_t (1 - \pi) l_t.$$

The first-order conditions are

$$c_t^i = c_t^s,$$

$$\eta \frac{c_t^s}{1 - l_t} = w_t,$$

$$r_t - \delta' \left(u_t \right) = 0,$$

and

$$1 = E_t \left[\frac{\beta}{e^{d_{t+1}-d_t}} \left(\frac{c_t^i}{c_{t+1}^i} \right) \left(u_{t+1}r_{t+1} + 1 - \delta\left(u_{t+1} \right) \right) \right].$$

A.2 With Tight Financial Constraints

This section considers the household's problem when financial constraints are tight. We derive the feasibility constraint for investment,

$$(1 - \phi q_t) i_t = u_t r_t n_t + \phi q_t (1 - \delta (u_t)) n_t + \mathbf{1}_{\{z_t = b\}} \tilde{p}_t \tilde{m}_t.$$

We first show that $x_t^i = 0$ if $q_t > 1$. Suppose that $x_t^i > 0$ holds even though $q_t > 1$. Then, the household can increase its utility by increasing i_t by Δ , increasing n_{t+1}^i by $(1 - \phi) \Delta$, decreasing x_t^i by $(1 - \phi q_t) \Delta$, decreasing n_{t+1}^s by $\frac{\pi}{1-\pi} (1 - \phi) \Delta$, increasing x_t^s by $\frac{\pi}{1-\pi} (1 - \phi) q_t \Delta$, and increasing both c_t^i and c_t^s by $\pi (q_t - 1) \Delta$ for sufficiently small $\Delta > 0$. This is a contradiction to an equilibrium condition requiring that the household maximize its utility subject to the constraints.

Similarly, suppose that $\mathbf{1}_{\{z_t=b\}}\tilde{p}_t\tilde{m}_{t+1}^i > 0$ holds in some t even though $q_t > 1$. Then, the household can increase its utility by increasing i_t by Δ , increasing n_{t+1}^i by $(1-\phi)\Delta$, decreasing \tilde{m}_{t+1}^i by $\frac{1}{\tilde{p}_t}(1-\phi q_t)\Delta$, decreasing n_{t+1}^s by $\frac{\pi}{1-\pi}(1-\phi)\Delta$, increasing \tilde{m}_{t+1}^s by $\frac{\pi}{1-\pi}\frac{1}{\tilde{p}_t}(1-\phi q_t)\Delta$, and increasing both x_t^s and c_t^s by $\frac{\pi}{1-\pi}(q_t-1)\Delta$ for sufficiently small $\Delta > 0$. This is a contradiction to an equilibrium condition requiring that the household maximize its utility subject to the constraints.

With these observations, we can rewrite the investor's budget constraint

$$x_{t}^{i} + i_{t} + q_{t} \left(n_{t+1}^{i} - i_{t} - (1 - \delta \left(u_{t} \right)) n_{t} \right) + \mathbf{1}_{\{z_{t}=b\}} \tilde{p}_{t} \left(\tilde{m}_{t+1}^{i} - \tilde{m}_{t} \right) = u_{t} r_{t} n_{t}$$

and obtain the feasibility constraint for investment.

B Full Model

This section presents the full model.

B.1 Micro-Foundations of Financial Frictions

We describe the micro-foundations of financial frictions. Because they are related to the household, we describe its problem in detail.

The economy is populated by a continuum of households, with measure one. All households behave identically. Each household has a unit measure of members who are identical at the beginning of each period. During the period, members are separated from each other, and each member receives a shock that determines her role in the period. A member will be an investor with probability $\pi \in [0, 1]$ and will be a saver/worker with probability $1 - \pi$. These shocks are i.i.d. among members and across time.

A period is divided into three stages. In the first stage, all members of a household are together and pool their assets, which are holdings of capital and, if it is the bubbly regime, holdings of bubbly assets. Aggregate shocks to exogenous state variables are realized. The household decides how intensively to use the capital it owns (i.e., the capacity utilization rate). Because all the members of the household are identical in this stage, the household head evenly divides the assets among the members. The household head also gives contingency plans to each member, describing the actions a member should take if she becomes an investor or a saver/worker.

At the beginning of the second stage, each member receives the shock determining her role in the period. Markets open and competitive firms produce final goods. Compensation for productive factors is paid to their owners. A fraction of capital depreciates. Investors seek financing to undertake investment projects. They have the technology to transform any amount of final goods into the same amount of new capital.

Following Kiyotaki and Moore (2012), we assume that investors face a borrowing constraint due to their lack of commitment power. Namely, an investor who produces new capital cannot fully precommit to work with it even though her specific skill will be needed for capital to provide services. As a result, an investor can only issue new equity up to a fraction θ of her investment. Specifically, the inequality constraint

$$issue_t \le \theta i_t$$
 (24)

must be satisfied, where i_t denotes the amount of new capital produced by an investor and $issue_t$ denotes the amount of equity issued by the same investor. The rest of the new capital cannot be sold due to the lack of the commitment power. It must be held privately.

Investors however can still use privately held capital as collateral to borrow short-term funds. Specifically, investors can choose the amount of borrowing $loan_t^i$, but it has to satisfy the inequality $\operatorname{constraint}$

$$loan_t^i \le \tilde{\phi}_t \left(1 - \delta\left(u_t\right)\right) n_{p,t} \tag{25}$$

where $n_{p,t}$ is the amount of privately held capital the investor has and $\tilde{\phi}_t$ is a time-varying parameter. If $loan_t^i$ is negative, the investor is a lender. As we explain momentarily, loans are repaid from the household's budget in the consumption stage.

Investors have equity issued by other households in their portfolio. They can sell it in the stock market, but there is a limit to this activity. Specifically, following Kiyotaki and Moore (2012), we assume that an investor can sell a fraction $\phi < 1$ of her holdings of other households' equity before the investment opportunity disappears. This is equivalent to introducing transaction costs that are zero for the first fraction ϕ of equity sold, and then infinite. Let $n_{e,t}$ and $n_{e,t+1}^i$ denote the investor's holding of other households' equity at the beginning and at the end of the investment stage, respectively. The resalability constraint is given by

$$n_{e,t+1}^{i} \ge (1 - \phi) (1 - \delta (u_t)) n_{e,t}.$$
(26)

Finally, investors have bubbly assets in their portfolio if the economy is in the bubbly regime. They can sell them freely in the bubbly regime. In the fundamental regime, there are neither spot nor future markets for bubbly assets. Without markets, no one can purchase bubbly assets, which is formally stated as follows:

$$\mathbf{1}_{\{z_t=f\}}\tilde{m}_{t+1}^i = \mathbf{1}_{\{z_t=f\}}\tilde{m}_{t+1}^s = 0.$$
(27)

Our assumptions about asset tradings lead to the following flow budget constraint of investors:

$$\underbrace{x_t^i + i_t + \underbrace{q_t \left(n_{e,t+1}^i - (1 - \delta\left(u_t\right)\right) n_{e,t}\right)}_{\text{net equity purchase}} + \underbrace{\mathbf{1}_{\{z_t=b\}} \tilde{p}_t \left(\tilde{m}_{t+1}^i - \tilde{m}_t\right)}_{\text{net bubble purchase}} = \underbrace{u_t r_t \left(n_{e,t} + n_{p,t}\right)}_{\text{dividend}} + \underbrace{q_t \left(issue_t\right)}_{\text{equity finance}} + loan_t^i,$$

$$\underbrace{u_t r_t \left(n_{e,t} + n_{p,t}\right)}_{\text{income + borrowing}} + \underbrace{q_t \left(issue_t\right)}_{\text{income + borrowing}} + \underbrace{q$$

The saver's flow budget constraint is similar to the investor's:

$$\underbrace{x_t^s + q_t \left(n_{e,t+1}^s - (1 - \delta\left(u_t\right)\right) n_{e,t}\right) + \mathbf{1}_{\{z_t=b\}} \tilde{p}_t \left(\tilde{m}_{t+1}^s - \tilde{m}_t\right)}_{\text{spending}} = \underbrace{u_t r_t \left(n_{e,t} + n_{p,t}\right) + w_t l_t + loan_t^s}_{\text{income + lending}}.$$
(29)

Here, x_t^s , $n_{e,t+1}^s$, \tilde{m}_{t+1}^s , and $loan_t^s$ are saver's counterparts of x_t^i , $n_{e,t+1}^i$, \tilde{m}_{t+1}^i , and $loan_t^i$ in equation (28). Savers also face the same constraints regarding asset tradings as investors. But we omit them because they do not bind in equilibrium.

The members of the household get together in the consumption stage. The short-term loans are paid back from the household's budget. In a symmetric equilibrium,

$$\pi loan_t^i + (1 - \pi) \, loan_t^s = 0$$

holds. Then, consumption takes place. The household's resource constraint at this point is

$$\pi x_t^i + (1 - \pi) x_t^s = \pi c_t^i + (1 - \pi) c_t^s.$$
(30)

After consumption, members' identities are lost. They start a new period as identical members. The household's portfolio at the beginning of period t + 1 consists of holdings of other households' equity given by

$$n_{e,t+1} = \pi n_{e,t+1}^{i} + (1 - \pi) n_{e,t+1}^{s}, \qquad (31)$$

privately held capital given by

$$n_{p,t+1} = (1 - \delta(u_t)) n_{p,t} + \pi (i_t - issue_t), \qquad (32)$$

and bubbly assets given by

$$\tilde{m}_{t+1} = \pi \tilde{m}_{t+1}^i + (1 - \pi) \,\tilde{m}_{t+1}^s + \mathbf{1}_{\{z_t = f, z_{t+1} = b\}} M.$$
(33)

The household's problem is summarized as follows. It chooses a sequence of $u_t, x_t^i, c_t^i, i_t, n_{e,t+1}^i, \tilde{m}_{t+1}^i, loan_t^i, issue_t, x_t^s, c_t^s, l_t, n_{e,t+1}^s, \tilde{m}_{t+1}^s$, and $loan_t^s$ to maximize the utility function

$$E_0\left[\sum_{t=0}^{\infty} \frac{\beta^t}{e^{d_t}} \left(\pi \log\left(c_t^i\right) + (1-\pi)\left[\log\left(c_t^s\right) + \eta\left(1-l_t\right)\right]\right)\right]$$

subject to the constraints (24), (25), (26), (27), (28), (29), (30), (31), (32), and (33). The initial portfolio $\{n_{e,0}, n_{p,0}, \tilde{m}_0\}$ is given. Except for $loan_t^i$ and $loan_t^s$, the control variables must be non-negative.

B.2 Market Clearing Conditions

We have introduced new assets, i.e., equity and privately held capital. There is no market for the privately held capital because no one can sell it due to the lack of commitment power. The equity market clearing condition is given by

$$n_{e,t+1} = (1 - \delta(u_t)) n_{e,t} + \pi(issue_t).$$

Market clearing conditions for labor services, capital services, and final goods are the same as in the baseline model, and so is the market clearing condition for the bubbly assets in the bubbly regime. In addition, the consistency condition

$$n_{e,t} + n_{p,t} = K_t$$

is satisfied for all t.

B.3 Simplifying Assumptions

Because the aforementioned problem is hard to analyze in the general form, we make a simplifying assumption following Kiyotaki and Moore (2012) and Del Negro et al. (2017). Specifically, we assume that $\tilde{\phi}_t = \phi q_t$ always holds. It can be justified in several ways. For example, if lenders can convert a unit of uncommitted capital into ϕ units of general capital that can be easily used by anyone and hence sold in the equity market, $\tilde{\phi}_t = \phi q_t$ holds.

With this assumption, the household no longer has to keep track of these two assets separately, but the total capital it owns, $n_t \equiv n_{e,t} + n_{p,t}$, becomes the relevant state variable for the household. This is because the other households' equity and the household's privately held capital become perfect substitutes for the household, paying the same return per unit and providing the same amount of liquidity per period. q_t is not only the equity price but also the household's subjective valuation of privately held capital. Finally, following Kiyotaki and Moore (2012), we assume that $\theta = \phi$ holds to simplify the analysis. The full model is now effectively the same as the original model. But the distinction between $n_{e,t}$ and $n_{p,t}$ is still important for the measurement of the stock market value.

B.4 Stock Market Value

We assume that $n_{e,0} = \phi K_0$ holds in period 0, which implies that $n_{e,t+1} = \phi K_{t+1}$ holds for $t \ge 0$ too.³¹ The stock market value is then given by

$$stock_t = \phi \left[q_t K_{t+1} \right] + \tilde{p}_t \mathbf{1}_{\{z_t=b\}} M.$$

This is identical to the stock market value we gave in the main text.

We close this section by discussing a caveat. Our assumption about uncommitted capital is slightly different from Kiyotaki and Moore's (2012). That is, while we assume that investors use it as collateral to borrow funds, they assume that investors gain additional commitment power to the uncommitted old capital every period and sell it in the equity market up to a certain limit. Our model behaves identically under their assumption except for the stock market value. Specifically, the equity-to-capital ratio has history dependence under their assumption, and we have to keep track of this ratio as an endogenous state variable in the estimation. This is technically demanding for our study, because our model has regime switches. Our assumption that investors borrow shortterm funds avoids this issue because it makes the equity-to-capital ratio constant, simplifying the analysis.

 $[\]overline{{}^{31}issue_t = \phi i_t}$ is optimal if $q_t > 1$ holds. If $q_t = 1$ holds, any level of equity issuance between 0 and ϕi_t is optimal, and we assume that they choose $issue_t = \phi i_t$.

C Model Summary

C.1 Fundamental Equilibrium With Loose Financial Constraints

When financial constraints are sufficiently loose, the equilibrium conditions are summarized as follows:

$$\begin{split} Y_t &= \bar{A}e^{a_t} u_t^{\alpha} K_t \left(\left(1 - \pi \right) l_t \right)^{1 - \alpha}, \\ \eta \frac{c_t}{1 - l_t} &= w_t, \\ \delta' \left(u_t \right) &= r_t, \\ 1 &= E_t \left[\frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{c_t}{c_{t+1}} \right) \left(u_{t+1} r_{t+1} + 1 - \delta \left(u_{t+1} \right) \right) \right], \\ r_t &= \alpha \frac{Y_t}{u_t K_t}, \\ w_t &= \left(1 - \alpha \right) \frac{Y_t}{\left(1 - \pi \right) l_t}, \end{split}$$

and

$$c_t + K_{t+1} - (1 - \delta(u_t)) K_t = Y_t.$$

Detrending variables by K_t , we obtain

$$\begin{split} \hat{Y}_t &= \bar{A}e^{a_t}u_t^{\alpha}\left(\left(1-\pi\right)l_t\right)^{1-\alpha},\\ \eta \frac{\hat{c}_t}{1-l_t} &= \hat{w}_t,\\ \delta'\left(u_t\right) &= r_t,\\ 1 &= E_t\left[\frac{\beta}{e^{d_{t+1}-d_t}}\left(\frac{\hat{c}_t}{\hat{c}_{t+1}}\frac{1}{g_t}\right)\left(u_{t+1}r_{t+1}+1-\delta\left(u_{t+1}\right)\right)\right],\\ r_t &= \alpha \frac{\hat{Y}_t}{u_t},\\ \hat{w}_t &= (1-\alpha)\,\frac{\hat{Y}_t}{(1-\pi)\,l_t}, \end{split}$$

and

$$\hat{c}_t + g_t - (1 - \delta\left(u_t\right)) = \hat{Y}_t$$

where variables with a hat denote the original variables divided by K_t , for example, $\hat{Y}_t \equiv Y_t/K_t$.

C.2 Fundamental Equilibrium With Tight Financial Constraints

Suppose that the financial constraints are sufficiently tight that they are always binding. In addition, suppose that the economy is in the fundamental equilibrium. The equilibrium conditions are summarized as follows:

$$\begin{split} Y_t &= \bar{A}e^{a_t} u_t^{\alpha} K_t \left(\left(1 - \pi \right) l_t \right)^{1 - \alpha}, \\ \eta \frac{c_t}{1 - l_t} &= w_t, \\ r_t - \delta' \left(u_t \right) q_t + \pi \lambda_t \left(r_t - \phi q_t \delta' \left(u_t \right) \right) &= 0, \\ q_t &= E_t \left[\frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{c_t}{c_{t+1}} \right) \left(u_{t+1} r_{t+1} + \left(1 - \delta \left(u_{t+1} \right) \right) q_{t+1} + \pi \lambda_{t+1} \left(u_{t+1} r_{t+1} + \phi q_{t+1} \left(1 - \delta \left(u_{t+1} \right) \right) \right) \right) \right], \\ r_t &= \alpha \frac{Y_t}{u_t K_t}, \\ w_t &= \left(1 - \alpha \right) \frac{Y_t}{\left(1 - \pi \right) l_t}, \\ Y_t &= c_t + \pi \frac{\left[u_t r_t + \phi q_t \left(1 - \delta \left(u_t \right) \right) \right] K_t}{1 - \phi q_t}, \\ K_{t+1} &= \left(1 - \delta \left(u_t \right) \right) K_t + \pi \frac{\left[u_t r_t + \phi q_t \left(1 - \delta \left(u_t \right) \right) \right] K_t}{1 - \phi q_t}, \end{split}$$

and

$$\lambda_t = \frac{q_t - 1}{1 - \phi q_t}.$$

Detrending variables by K_t , we obtain

$$\begin{split} \hat{Y}_{t} &= \bar{A}e^{a_{t}}u_{t}^{\alpha}\left((1-\pi)\,l_{t}\right)^{1-\alpha},\\ &\eta\frac{\hat{c}_{t}}{1-l_{t}} = \hat{w}_{t},\\ &r_{t} - \delta'\left(u_{t}\right)q_{t} + \pi\lambda_{t}\left(r_{t} - \phi q_{t}\delta'\left(u_{t}\right)\right) = 0,\\ q_{t} &= E_{t}\left[\frac{\beta}{e^{d_{t+1}-d_{t}}}\left(\frac{\hat{c}_{t}}{\hat{c}_{t+1}}\frac{1}{g_{t}}\right)\left(u_{t+1}r_{t+1} + (1-\delta\left(u_{t+1}\right)\right)q_{t+1} + \pi\lambda_{t+1}\left(u_{t+1}r_{t+1} + \phi q_{t+1}\left(1-\delta\left(u_{t+1}\right)\right)\right)\right)\right],\\ &r_{t} = \alpha\frac{\hat{Y}_{t}}{u_{t}},\\ &\hat{w}_{t} = (1-\alpha)\frac{\hat{Y}_{t}}{(1-\pi)\,l_{t}},\\ &\hat{Y}_{t} = \hat{c}_{t} + \pi\frac{u_{t}r_{t} + \phi q_{t}\left(1-\delta\left(u_{t}\right)\right)}{1-\phi q_{t}}, \end{split}$$

$$g_{t} = 1 - \delta(u_{t}) + \pi \frac{u_{t}r_{t} + \phi q_{t}(1 - \delta(u_{t}))}{1 - \phi q_{t}},$$

$$\lambda_t = \frac{q_t - 1}{1 - \phi q_t}.$$

C.3 Recurrent-Bubble Equilibrium

Suppose that the economy is in the recurrent-bubble equilibrium. The equilibrium conditions are summarized as follows:

$$\begin{split} Y_t &= \bar{A} e^{a_t} u_t^{\alpha} K_t \left(\left(1 - \pi \right) l_t \right)^{1 - \alpha}, \\ & \eta \frac{c_t}{1 - l_t} = w_t, \\ r_t - \delta' \left(u_t \right) q_t + \pi \lambda_t \left(r_t - \phi q_t \delta' \left(u_t \right) \right) = 0, \\ q_t &= E_t \left[\frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{c_t}{c_{t+1}} \right) \left(u_{t+1} r_{t+1} + \left(1 - \delta \left(u_{t+1} \right) \right) q_{t+1} + \pi \lambda_{t+1} \left(u_{t+1} r_{t+1} + \phi q_{t+1} \left(1 - \delta \left(u_{t+1} \right) \right) \right) \right) \right], \\ \mathbf{1}_{\{z_t = b\}} \tilde{p}_t = \mathbf{1}_{\{z_t = b\}} E_t \left[\frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{c_t}{c_{t+1}} \right) \left(1 + \pi \lambda_{t+1} \right) \tilde{p}_{t+1} \mathbf{1}_{\{z_{t+1} = b\}} \right], \\ r_t &= \alpha \frac{Y_t}{u_t K_t}, \\ w_t &= \left(1 - \alpha \right) \frac{Y_t}{\left(1 - \pi \right) l_t}, \\ Y_t &= c_t + \pi \frac{\left[u_t r_t + \phi q_t \left(1 - \delta \left(u_t \right) \right) \right] K_t + \tilde{p}_t \mathbf{1}_{\{z_t = b\}} M}{1 - \phi q_t}, \\ K_{t+1} &= \left(1 - \delta \left(u_t \right) \right) K_t + \pi \frac{\left[u_t r_t + \phi q_t \left(1 - \delta \left(u_t \right) \right) \right] K_t + \tilde{p}_t \mathbf{1}_{\{z_t = b\}} M}{1 - \phi q_t}, \end{split}$$

and

$$\lambda_t = \frac{q_t - 1}{1 - \phi q_t}.$$

Detrending variables by K_t , we obtain

$$\begin{split} \hat{Y}_t &= \bar{A}e^{a_t}u_t^{\alpha}\left((1-\pi)\,l_t\right)^{1-\alpha},\\ \eta \frac{\hat{c}_t}{1-l_t} &= \hat{w}_t,\\ r_t - \delta'\left(u_t\right)q_t + \pi\lambda_t\left(r_t - \phi q_t\delta'\left(u_t\right)\right) &= 0, \end{split}$$

$$q_t &= E_t\left[\frac{\beta}{e^{d_{t+1}-d_t}}\left(\frac{\hat{c}_t}{\hat{c}_{t+1}}\frac{1}{g_t}\right)\left(u_{t+1}r_{t+1} + (1-\delta\left(u_{t+1}\right)\right)q_{t+1} + \pi\lambda_{t+1}\left(u_{t+1}r_{t+1} + \phi q_{t+1}\left(1-\delta\left(u_{t+1}\right)\right)\right)\right)\right],\\ m_t &= \mathbf{1}_{\{z_t=b\}}E_t\left[\frac{\beta}{e^{d_{t+1}-d_t}}\left(\frac{\hat{c}_t}{\hat{c}_{t+1}}\frac{1}{g_t}\right)\left(1+\pi\lambda_{t+1}\right)m_{t+1}g_t\right],\end{split}$$

$$\begin{split} r_t &= \alpha \frac{\hat{Y}_t}{u_t}, \\ \hat{w}_t &= (1-\alpha) \, \frac{\hat{Y}_t}{(1-\pi) \, l_t}, \\ \hat{Y}_t &= \hat{c}_t + \pi \frac{u_t r_t + \phi q_t \, (1-\delta \, (u_t)) + m_t}{1-\phi q_t}, \\ g_t &= 1-\delta \, (u_t) + \pi \frac{u_t r_t + \phi q_t \, (1-\delta \, (u_t)) + m_t}{1-\phi q_t}, \end{split}$$

$$\lambda_t = \frac{q_t - 1}{1 - \phi q_t}$$

where $m_t \equiv \tilde{p}_t \mathbf{1}_{\{z_t=b\}} M/K_t$. It is important that the system of equations summarized above does not have endogenous state variables. The endogenous variables in the system are therefore determined by exogenous state variables $\{z_t, a_t, d_t\}$.

It is convenient to make the regime-dependence explicit:

$$\hat{Y}_{f,t} = \bar{A}e^{a_t} \left(u_{f,t} \right)^{\alpha} \left(\left(1 - \pi \right) l_{f,t} \right)^{1-\alpha},$$
(34)

$$\hat{Y}_{b,t} = \bar{A}e^{a_t} \left(u_{b,t} \right)^{\alpha} \left(\left(1 - \pi \right) l_{b,t} \right)^{1-\alpha}, \tag{35}$$

$$\eta \frac{\hat{c}_{f,t}}{1 - l_{f,t}} = \hat{w}_{f,t},\tag{36}$$

$$\eta \frac{\hat{c}_{b,t}}{1 - l_{b,t}} = \hat{w}_{b,t},\tag{37}$$

$$r_{f,t} - \delta'(u_{f,t}) q_{f,t} + \pi \lambda_{f,t} \left(r_{f,t} - \phi q_{f,t} \delta'(u_{f,t}) \right) = 0,$$
(38)

$$r_{b,t} - \delta'(u_{b,t}) q_{b,t} + \pi \lambda_{b,t} \left(r_{b,t} - \phi q_{b,t} \delta'(u_{b,t}) \right) = 0,$$
(39)

$$q_{f,t} = E_t \left[(1 - \sigma_f) \frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{\hat{c}_{f,t}}{\hat{c}_{f,t+1}} \frac{1}{g_{f,t}} \right) \\ \left(u_{f,t+1} r_{f,t+1} + (1 - \delta \left(u_{f,t+1} \right)) q_{f,t+1} + \pi \lambda_{f,t+1} \left(u_{f,t+1} r_{f,t+1} + \phi q_{f,t+1} \left(1 - \delta \left(u_{f,t+1} \right) \right) \right) \right) \\ + \sigma_f \frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{\hat{c}_{f,t}}{\hat{c}_{b,t+1}} \frac{1}{g_{f,t}} \right) \\ \left(u_{b,t+1} r_{b,t+1} + (1 - \delta \left(u_{b,t+1} \right)) q_{b,t+1} + \pi \lambda_{b,t+1} \left(u_{b,t+1} r_{b,t+1} + \phi q_{b,t+1} \left(1 - \delta \left(u_{b,t+1} \right) \right) \right) \right) \right], \quad (40)$$

$$q_{b,t} = E_t \left[(1 - \sigma_b) \frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{\hat{c}_{b,t}}{\hat{c}_{b,t+1}} \frac{1}{g_{b,t}} \right) \\ \left(u_{b,t+1} r_{b,t+1} + (1 - \delta \left(u_{b,t+1} \right)) q_{b,t+1} + \pi \lambda_{b,t+1} \left(u_{b,t+1} r_{b,t+1} + \phi q_{b,t+1} \left(1 - \delta \left(u_{b,t+1} \right) \right) \right) \right) \\ + \sigma_b \frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{\hat{c}_{b,t}}{\hat{c}_{f,t+1}} \frac{1}{g_{b,t}} \right) \\ \left(u_{f,t+1} r_{f,t+1} + (1 - \delta \left(u_{f,t+1} \right)) q_{f,t+1} + \pi \lambda_{f,t+1} \left(u_{f,t+1} r_{f,t+1} + \phi q_{f,t+1} \left(1 - \delta \left(u_{f,t+1} \right) \right) \right) \right) \right], \quad (41)$$

$$m_{f,t} = 0, (42)$$

$$m_{b,t} = E_t \left[(1 - \sigma_b) \frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{\hat{c}_{b,t}}{\hat{c}_{b,t+1}} \frac{1}{g_{b,t}} \right) (1 + \pi \lambda_{b,t+1}) m_{b,t+1} g_{b,t} + \sigma_b \frac{\beta}{e^{d_{t+1} - d_t}} \left(\frac{\hat{c}_{b,t}}{\hat{c}_{f,t+1}} \frac{1}{g_{b,t}} \right) (1 + \pi \lambda_{f,t+1}) m_{f,t+1} g_{b,t} \right], \quad (43)$$

$$r_{f,t} = \alpha \frac{\hat{Y}_{f,t}}{u_{f,t}},\tag{44}$$

$$r_{b,t} = \alpha \frac{\hat{Y}_{b,t}}{u_{b,t}},\tag{45}$$

$$\hat{w}_{f,t} = (1-\alpha) \,\frac{\hat{Y}_{f,t}}{(1-\pi) \, l_{f,t}},\tag{46}$$

$$\hat{w}_{b,t} = (1 - \alpha) \, \frac{\hat{Y}_{b,t}}{(1 - \pi) \, l_{b,t}},\tag{47}$$

$$\hat{Y}_{f,t} = \hat{c}_{f,t} + \pi \frac{u_{f,t}r_{f,t} + \phi q_{f,t} \left(1 - \delta \left(u_{f,t}\right)\right) + m_{f,t}}{1 - \phi q_{f,t}},$$
(48)

$$\hat{Y}_{b,t} = \hat{c}_{b,t} + \pi \frac{u_{b,t} r_{b,t} + \phi q_{b,t} \left(1 - \delta \left(u_{b,t}\right)\right) + m_{b,t}}{1 - \phi q_{b,t}},\tag{49}$$

$$g_{f,t} = 1 - \delta\left(u_{f,t}\right) + \pi \frac{u_{f,t}r_{f,t} + \phi q_{f,t}\left(1 - \delta\left(u_{f,t}\right)\right) + m_{f,t}}{1 - \phi q_{f,t}},\tag{50}$$

$$g_{b,t} = 1 - \delta\left(u_{b,t}\right) + \pi \frac{u_{b,t}r_{b,t} + \phi q_{b,t}\left(1 - \delta\left(u_{b,t}\right)\right) + m_{b,t}}{1 - \phi q_{b,t}},\tag{51}$$

$$\lambda_{f,t} = \frac{q_{f,t} - 1}{1 - \phi q_{f,t}},\tag{52}$$

$$\lambda_{b,t} = \frac{q_{b,t} - 1}{1 - \phi q_{b,t}} \tag{53}$$

where subscripts f and b denote realizations of the variables in the fundamental and bubbly regimes, respectively; for instance, $\hat{Y}_{f,t}$ is the realization of \hat{Y}_t in the fundamental regime. The regime-dependent steady states are obtained as the solutions of the system of nonlinear equations (34) to (53) under the assumption that both a_t and d_t are constant at zero. To capture the effects of a_t and d_t , we linearize the equations (34) to (53) around the regime-dependent steady states and obtain the impulse response functions.

D Solving the Simple Model

This section shows how to solve the simple model considered in the paper.

D.1 After Bubbly Episode

The Euler equation for capital, equation (40), is written as

$$q_f^* = \beta \frac{r}{g_f^*} \left(1 + \pi \frac{q_f^* - 1}{1 - \phi q_f^*} \right).$$
(54)

The feasibility constraint for investment, equation (50), is written as

$$g_f^* = \pi \frac{r}{1 - \phi q_f^*}.$$
 (55)

Substituting (55) into (54) and rearranging terms, we obtain

$$q_f^* = \frac{\beta \left(1 - \pi\right)}{\pi \left(1 - \beta\right) + \beta \phi}.$$
(56)

Substituting it into (55) and rearranging terms, we obtain

$$g_{f}^{*} = r \left[1 - \frac{(1-\beta)(1-\pi)}{1-\beta+\beta\phi} \right].$$
 (57)

D.2 During Bubbly Episode

The Euler equation for bubbly assets, equation (43), is written as

$$m_b = (1 - \sigma_b) \beta \left(1 + \pi \frac{q_b - 1}{1 - \phi q_b} \right) m_b.$$
(58)

The Euler equation for capital, equation (41), is written as

$$q_b = \frac{r}{g_b} \left[\left(1 - \sigma_b\right) \beta \left(1 + \pi \frac{q_b - 1}{1 - \phi q_b}\right) + \sigma_b \beta \left(\frac{\hat{c}_b}{\hat{c}_f^*}\right) \left(1 + \pi \frac{q_f^* - 1}{1 - \phi q_f^*}\right) \right].$$
(59)

From equations (48), (50), and $\hat{Y}_f = r$, we obtain $\hat{c}_f^* = r - g_f^*$. Similarly, from equations (49), (51), and $\hat{Y}_b = r$, we obtain $\hat{c}_b = r - g_b$. The feasibility constraint for investment, equation (51), is written as

$$g_b = \pi \frac{r + m_b}{1 - \phi q_b}.\tag{60}$$

We solve equations (58), (59), and (60) for m_b , q_b , and g_b . $\{m_b, q_b, g_b\} = \{0, q_f^*, g_f^*\}$ solve these equations. But we are not interested in this bubbleless equilibrium now. So we assume that $m_b \neq 0$ holds.

Dividing both sides of (58) by m_b and rearranging terms, we obtain

$$q_b = \frac{(1-\pi)\beta + (\sigma_b - \bar{\sigma}_b)\frac{\beta(1-\pi)^2}{1-\phi}}{\pi(1-\beta) + \beta\phi + (\sigma_b - \bar{\sigma}_b)\frac{\beta(\phi-\pi)(1-\pi)}{1-\phi}},$$
(61)

where $\bar{\sigma}_b \equiv \frac{1}{1-\pi} \left[2 - \pi - \frac{1}{\beta} - \phi \right]$. Using (54) and (58), we simplify (59) as follows;

$$q_b = \frac{r}{g_b} \left[1 + \sigma_b \left(\frac{\hat{c}_b}{\hat{c}_f^*} \right) \frac{q_f^* g_f^*}{r} \right].$$

Substituting $\hat{c}_b = r - g_b$ and (60) into this equation and rearranging terms, we obtain

$$q_b \pi \left(1 + \frac{m_b}{r}\right) = 1 - \phi q_b + \sigma_b \left(\frac{q_f^* g_f^*}{\hat{c}_f^*}\right) \left(1 - \phi q_b - \pi \left(1 + \frac{m_b}{r}\right)\right).$$

Substituting $\hat{c}_f^* = r - g_f^*$, (56), and (57) into this equation and rearranging terms, we obtain

$$m_b = \frac{r\beta}{\beta\pi + 1 - \beta + \sigma_b\beta\left(\phi - \pi\right)} \left[2 - \pi - \frac{1}{\beta} - \left(\phi + \sigma_b\left(1 - \pi\right)\right)\right].$$
(62)

Substituting (61) and (62) into (60) and rearranging terms, we obtain

$$g_b = r \left[1 - \frac{(1-\beta)(1-\pi)}{1-\beta+\beta\phi+(\sigma_b-\bar{\sigma}_b)\frac{\beta(1-\pi)(\phi-\pi)}{1-\phi}} \right].$$
 (63)

D.3 Crowding-In Effect of Realized Bubble

We show that the size of the bubble relative to the capital stock, m_b , decreases with the level of the financial development, ϕ . Given the analytical solution (62), we can directly prove it by taking a derivative. Namely, we have

$$\frac{\partial m_b}{\partial \phi} = -\left(\frac{1}{\beta \pi + 1 - \beta + \sigma_b \beta \left(\phi - \pi\right)}\right)^2 r\beta \left(1 - \sigma_b\right) \left[1 - \beta \left(1 - \pi\right) \left(1 - \sigma_b\right)\right] < 0.$$

D.4 Crowding-Out Effect of Realized Bubble

We show that the price of capital during the bubbly episode, q_b , is smaller than the price of capital after the bubbly episode, q_f^* . It can be shown in two steps. First, notice that q_b converges to q_f^* as σ_b converges to $\bar{\sigma}_b$, namely, $\lim_{\sigma_b \nearrow \bar{\sigma}_b} q_b = q_f^*$. This is obvious from the analytical solutions (56) and (61). Second, taking a derivative of q_b with respect to σ_b , we obtain

$$\frac{\partial q_b}{\partial \sigma_b} = \left(\frac{1-\pi}{\pi \left(1-\beta\right) + \beta \phi + \left(\sigma_b - \bar{\sigma}_b\right) \frac{\beta(\phi-\pi)(1-\pi)}{1-\phi}}\right)^2 \frac{\beta \pi}{1-\phi} > 0.$$

Hence, q_b is increasing in σ_b and converges to q_f^* as σ_b converges to $\bar{\sigma}_b$. Because we assume that σ_b is less than $\bar{\sigma}_b$, q_b is smaller than q_f^* .

D.5 Proposition 1

We show Proposition 1. From equations (57) and (63), it is obvious that g_b converges to $\bar{\sigma}_b$, namely, $\lim_{\sigma_b \nearrow \bar{\sigma}_b} g_b = g_f^*$. In addition, g_b decreases with σ_b if and only if $\phi < \pi$, which is obvious from the analytical solution (63). Because we assume that σ_b is less than $\bar{\sigma}_b$, g_b is larger than g_f^* if and only if $\phi < \pi$.

D.6 Before Bubbly Episode

The Euler equation for capital, equation (40), is written as

$$q_f = \frac{r}{g_f} \left[\left(1 - \sigma_f\right) \beta \left(1 + \pi \frac{q_f - 1}{1 - \phi q_f}\right) + \sigma_f \beta \left(\frac{\hat{c}_f}{\hat{c}_b}\right) \left(1 + \pi \frac{q_b - 1}{1 - \phi q_b}\right) \right].$$
(64)

The feasibility constraint for investment, equation (50), is written as

$$g_f = \pi \frac{r}{1 - \phi q_f}.\tag{65}$$

We solve equations (64) and (65) for q_f and g_f . Using (58), we can simplify (64) as follows;

$$q_f = \frac{r}{g_f} \left[\left(1 - \sigma_f\right) \beta \left(1 + \pi \frac{q_f - 1}{1 - \phi q_f}\right) + \sigma_f \left(\frac{\hat{c}_f}{\hat{c}_b}\right) \frac{1}{1 - \sigma_b} \right].$$

Substituting $\hat{c}_f = r - g_f$ and (65) into this equation and rearranging terms, we obtain

$$q_f = \frac{\left(1 - \pi\right)\beta + \sigma_f\left(1 - \pi\right)\left(-\beta + \frac{1}{1 - \sigma_b}\frac{r}{\hat{c}_b}\right)}{\pi\left(1 - \beta\right) + \beta\phi + \sigma_f\left(\beta\left(\pi - \phi\right) + \frac{\phi}{1 - \sigma_b}\frac{r}{\hat{c}_b}\right)}.$$

Given q_f , capital growth g_f is determined by equation (65).

D.7 Proposition 3

We show Proposition 3. The stock market to GDP ratio before, during, and after the bubbly episode is, respectively, given by

$$\frac{stock_{t}}{4Y_{t}} = \begin{cases} \frac{\phi q_{f}g_{f}}{4r}, & \text{if } z_{\tau} = f \text{ and for all } \tau \leq t, \\ \frac{\phi q_{b}g_{b}+m_{b}}{4r} = \frac{\phi \beta (1-\pi) - (\sigma_{b}-\bar{\sigma}_{b})\beta (1-\pi)^{2}}{4\left(1-\beta(1-\phi)+(\sigma_{b}-\bar{\sigma}_{b})\frac{\beta(1-\pi)(\phi-\pi)}{1-\phi}\right)}, & \text{if } z_{t} = b, \\ \frac{\phi q_{f}^{*}g_{f}^{*}}{4r} = \frac{\phi \beta (1-\pi)}{4(1-\beta(1-\phi))}, & \text{otherwise.} \end{cases}$$

The stock market to GDP ratio during the bubbly episode decreases with σ_b because

$$\frac{\partial}{\partial \sigma_b} \left(\frac{\phi q_b g_b + m_b}{4r} \right) = -\frac{1}{4} \left(\frac{\beta \left(1 - \pi \right)}{1 - \beta \left(1 - \phi \right) + \left(\sigma_b - \bar{\sigma}_b \right) \frac{\beta \left(1 - \pi \right) \left(\phi - \pi \right)}{1 - \phi}} \right)^2 \left(\frac{1 - \beta}{\beta} + \frac{\phi \left(1 - \pi \right)}{1 - \phi} \right) < 0.$$

In addition, the ratio during the bubbly episode converges to the one after the bubbly episode as σ_b converges to $\bar{\sigma}_b$, namely,

$$\lim_{\sigma_b \nearrow \bar{\sigma}_b} \left(\frac{\phi q_b g_b + m_b}{4r} \right) = \frac{\phi q_f^* g_f^*}{4r}$$

Because we assume that σ_b is smaller than $\bar{\sigma}_b$, the stock market to GDP ratio during the bubbly episode is larger than the one after the bubbly episode. In addition, the stock market to GDP ratio after the bubbly episode is larger than the one before the bubbly episode because Proposition 2 implies that both $q_f < q_f^*$ and $g_f < g_f^*$ hold. Therefore, the stock market to GDP ratio during the bubbly episode is larger than the one before the bubbly episode too.

D.8 Comovement Problem

This section shows that the simple model suffers from a comovement problem. Investment growth is given by

$$\frac{i_t}{i_{t-1}} = \frac{K_{t+1}}{K_t} = g_t = \begin{cases} g_f, & \text{if } z_\tau = f \text{ and for all } \tau \le t, \\ g_b, & \text{if } z_t = b, \\ g_f^*, & \text{otherwise.} \end{cases}$$

Similarly, consumption growth is given by

$$\frac{c_t}{c_{t-1}} = \frac{\hat{c}_t}{\hat{c}_{t-1}} \frac{K_t}{K_{t-1}} = \frac{r - g_t}{r - g_{t-1}} g_{t-1} = \begin{cases} g_f, & \text{if } z_\tau = f \text{ and for all } \tau \le t, \\ \left(1 + \frac{g_f - g_b}{r - g_f}\right) g_f, & \text{if } \{z_{t-1}, z_t\} = \{f, b\}, \\ g_b, & \text{if } \{z_{t-1}, z_t\} = \{b, b\}, \\ \left(1 + \frac{g_b - g_f^*}{r - g_b}\right) g_b, & \text{if } \{z_{t-1}, z_t\} = \{b, f\}, \\ g_f^*, & \text{otherwise.} \end{cases}$$

If $g_f < g_b$ holds, investment growth rises from g_f to g_b when the bubble emerges (in period t with $\{z_{t-1}, z_t\} = \{f, b\}$) but consumption growth drops. Similarly, if $g_b > g_f^*$ holds, investment growth drops from g_b to g_f^* when the bubble bursts (in period t with $\{z_{t-1}, z_t\} = \{b, f\}$) but consumption growth rises. Clearly, it is impossible to have comovement between investment and consumption.

E Transversality Conditions

We show that the transversality conditions are satisfied along the balanced growth path. Let's assume without loss of generality that $a_t = d_t = 0$ for all $t \ge 0$. Then, we have

$$\left(\frac{1}{c_t^i}\right)q_t K_{t+1} = \left(\frac{1}{\hat{c}_t^i}\right)q_t g_t = \begin{cases} \left(\frac{1}{\hat{c}_f^i}\right)q_f g_f, & \text{if } z_t = f, \\ \left(\frac{1}{\hat{c}_b^i}\right)q_b g_b, & \text{if } z_t = b. \end{cases}$$

Similarly, we have

$$\left(\frac{1}{c_t^i}\right)\tilde{p}_t \mathbf{1}_{\{z_t=b\}} M = \left(\frac{1}{\hat{c}_t^i}\right) m_t = \begin{cases} 0, & \text{if } z_t = f, \\ \left(\frac{1}{\hat{c}_b^i}\right) m_b, & \text{if } z_t = b. \end{cases}$$

The transversality conditions are satisfied because

$$0 \le \lim_{t \to \infty} E_0 \left[\beta^t \left(\frac{1}{c_t^i} \right) q_t K_{t+1} \right] \le \lim_{t \to \infty} \beta^t \times \left[\max \left\{ \left(\frac{1}{\hat{c}_f^i} \right) q_f g_f, \left(\frac{1}{\hat{c}_b^i} \right) q_b g_b \right\} \right] = 0$$



Figure 11: Partial Collapse vs. Entire Collapse (Capital Growth)

$$0 \le \lim_{t \to \infty} \mathbf{1}_{\{z_0=b\}} E_0 \left[\beta^t \left(\frac{1}{c_t^i} \right) \tilde{p}_t \mathbf{1}_{\{z_t=b\}} M \right] \le \lim_{t \to \infty} \beta^t \left(\frac{1}{\hat{c}_b^i} \right) m_b = 0.$$

F Partial-Collapse Model

This section examines an alternative assumption replacing the fundamental regime with a lowbubble regime in which a small fraction of bubbly assets survive from the previous regime. This model has two bubbly regimes with different amounts of bubbly assets. We call them highbubble (H) and low-bubble (L) regimes respectively, in each of which M and $(1 - \delta_M) M$ units of bubbly assets exist respectively. A fraction $\delta_M \in (0, 1)$ of randomly chosen bubbly assets physically disappears when the regime switches to the low-bubble one, and $\delta_M M$ units of a new vintage of bubbly assets are created when the regime switches to the other direction. We omit the productivity and preference shocks to simplify the analysis.

Green circles and crosses in Figure 11 show the regime-dependent capital growth in this model. We set the depreciation rate of the bubbly asset at $\delta_M = 0.999$. Therefore, nearly all the bubbly assets disappear when the regime switches to the low-bubble regime. Nonetheless, the regime-dependent capital growth in the partial-collapse model does not resemble its counterpart in the original model plotted in red circles and crosses in the same figure. Specifically, the distance between green circles and crosses is a lot shorter than the distance between red circles and crosses.



Figure 12: Partial Collapse vs. Entire Collapse (Bubble Size)

Figure 12 explains why. It plots the regime-dependent bubble size relative to the capital stock. Importantly, a sizable bubble exists in the low-bubble regime. The mechanism is simple; even if most of the bubbly assets lose value (physically disappear in the model), the rest of the bubbly assets appreciate because liquid assets become scarce and demand for the remaining bubbly assets rises. This general equilibrium effect stabilizes the impact of the collapse. Our benchmark model is different in this respect; because we consider the entire collapse of bubbles as in Weil (1987), the supply of bubbly assets is zero in the fundamental regime, and therefore, the aforementioned general equilibrium effect is absent. As a consequence, the entire collapse of bubbles has a much stronger impact on growth.

Figure 13 plots the regime-dependent capital growth in this alternative model as a function of δ_M . We set $\phi = 0.15$, but the result is robust to other values of ϕ . At $\delta_M = 1$, we plot the regime-dependent capital growth in our benchmark model. We see no sign of "convergence" from the model with multiple partial collapses to the benchmark model as δ_M approaches 1. There is a discrete jump at $\delta_M = 1$. This is the same type of non-linearity that Brunnermeier and Sannikov (2014), He and Krishnamurthy (2013), and Gertler and Kiyotaki (2015) emphasize as an important factor to account for the financial crisis.



Figure 13: Partial Collapse vs. Entire Collapse (Depreciation and Growth)

G Data

In this section, we explain the observables used to estimate the model. The data consist of quarterly GDP growth and the stock-market-to-GDP ratio for the period 1984.Q1-2017.Q4. The data come from the St. Louis Fed's FRED database. For the stock-market-to-GDP ratio, we use the quarterly not seasonally adjusted Wilshire 5000 Full Cap Price Index series. The raw unfiltered series was used to compute GDP growth. We pre-filtered the stock market-to-GDP ratio series with the HP filter to remove the trend in the data that is not present in our model; see the main text for a discussion of the properties of the filtered series. This approach is reasonable because we are interested in understanding how the fluctuations around this trend are influenced by the presence or the absence of bubbles. Furthermore, this de-trending approach is standard in policy institutions such as the Federal Reserve System when it analyzes the evolution of credit in the economy (Bassett et al. (2015)). The Bank of Japan takes a similar approach too. The bank constructs the "heat map" from several financial indicators, including stock market value, on which abnormal deviations of a variable from the trend are read as a sign of over-heating. Please see the Financial System Report, a biannual publication of the bank surveying the financial system.³²

³²https://www.boj.or.jp/en/research/brp/fsr/index.htm/.

H Solution Method

The solution and estimation of the model requires a series of steps that we describe next.

- 1. We de-trend the model's equilibrium conditions by the stock of capital, resulting in a stationary model. It is easy to see that given the structural shocks and the regime today, the model is entirely forward looking (equations (34) to (53) in Section C.3).
- 2. Let X_t^f and Y_t^f denote the vectors containing the states and controls in the fundamental regime. Similarly, X_t^b and Y_t^b denote the vectors containing the states and controls in the bubbly regime. Then the de-trended model can be written as

$$E_t \Gamma_f(X_t^f, Y_t^f, X_{t+1}^f, Y_{t+1}^f, X_{t+1}^b, Y_{t+1}^b) = 0.$$

$$E_t \Gamma_b(X_t^b, Y_t^b, X_{t+1}^f, Y_{t+1}^f, X_{t+1}^b, Y_{t+1}^b) = 0.$$

That is, we stack the model's equilibrium equations conditional on being in the fundamental and the bubbly regimes. Note that the notation makes clear that the economy may switch to a different regime tomorrow. The functional equations describing the equilibrium conditions are captured by $\Gamma_f(\cdot)$ and $\Gamma_b(\cdot)$.

3. We compute the steady state (w/o structural shocks) of each regime (X^f, Y^f, X^b, Y^b) by shutting down the structural shocks but preserving the regime switches. In other words, we look for X^f, Y^f, X^b, Y^b that solve the system:

$$\Gamma_f(X^f, Y^f, X^f, Y^f, X^b, Y^b) = 0.$$

$$\Gamma_b(X^b, Y^b, X^f, Y^f, X^b, Y^b) = 0.$$

In doing so, our method respects the probability of switching from the fundamental steady state to the bubbly steady state and vice versa.

- 4. We perturb the model around the steady states and solve the resulting system to obtain the laws of motion for the endogenous states and controls. For simulations and estimation, we use a first-order perturbation approach (Schmitt-Grohe and Uribe, 2004).
- 5. It can be shown that the first-order approximation of the model can be written compactly as follows:

$$\mathbb{X}_t = \Lambda_x \mathbb{X}_{t-1} + \Omega_x \Xi_{x,t}.$$

Here, $\mathbb{X}_t = [X^f, Y^f, X^b, Y^b]'$ and $\Xi_{x,t}$ contains the structural innovations at time t.

6. We supplement the transition equation in the previous point with a measurement equation

of the form:

$$\mathbb{Y}_t = \Lambda_y \mathbb{X}_t + \Omega \Xi_{y,t}.$$

The matrix Λ_y makes the necessary transformations to make the model's variables compatible with the observables in the data collected in vector \mathbb{Y}_t . We allow for classical measurement errors as captured by \mathbb{Y}_t .

- 7. To compute the likelihood of the model, we use the nonlinear filter discussed in chapter 5 in Kim and Nelson (1999).
- 8. The Bayesian estimation is implemented following Fernandez-Villaverde et al. (2016).

I Impulse Responses

This section discusses the impulse response functions of variables not discussed in the paper. Table 3 reports responses to a one-standard-deviation innovation to a productivity shock $(SD_a = 0.01)$ and a preference shock $(SD_b = 0.08)$. Their auto-correlations are 0.9 and 0.5, respectively. We report contemporaneous responses on impact of the shock alone, because they are sufficient to summarize the impulse responses for the variables reported in the table. This is because all the variables in the table are determined by the exogenous state variables $\{z_t, a_t, d_t\}$ alone.

A positive productivity shock (a rise in a_t) increases output, consumption, investment, and hours worked simultaneously. In contrast, a positive preference shock (a rise in d_t) increases investment but decreases consumption. Remember that the preference shock decreases the level of the subjective discount factor on impact but it is mean reverting. Hence, after the shock, households end up assigning large weights to the utility flows in the distant future relative to those in the near future. Therefore, households become effectively more patient than before, hence increasing investment and decreasing consumption. Asset prices also increase because of the discount factor channel.

Comparing responses across regimes, we see larger responses in the bubbly regime than in the fundamental regime. Bubbles amplify the impact of the shocks because the bubble size positively responds to the shocks, supplying more liquidity to the economy. But the regime-dependence is relatively mild.

J Alternative Identification Strategies

In this section, we show the impact of alternative identification strategies on our empirical results. For our first check, we use quarterly U.S. data on GDP growth and the credit-to-GDP ratio. Similar to the stock market value, the credit-to-GDP ratio in the model is higher during bubbly episodes than during the fundamental ones. Figure 14 presents the estimated probability of the economy

	Bubbly Regime		Fundamental Regime	
	Productivity	Preference	Productivity	Preference
output-to-capital	1.18%	-0.00%	1.10%	-0.05%
consumption-to-capital	1.06%	-0.28%	1.03%	-0.26%
investment-to-capital	1.55%	0.84%	1.38%	0.87%
hours	0.09%	0.21%	0.05%	0.16%
utilization	0.36%	-0.42%	0.21%	-0.47%
capital price	0.77%	0.62%	0.95%	0.67%
bubble-to-capital	1.83%	0.79%	-	-
capital growth	0.05%	0.06%	0.04%	0.04%

Table 3: Effects of Productivity and Preference Shocks

being in the bubbly regime. It shows that the economy spent more time in the fundamental regime prior to the 2000s. This means that during the first 15 years of the sample, growth was driven by exogenous productivity shocks (not shown), not a surprise given the moderate credit-to-GDP ratio in the data.

The economy starts the 2000s in the fundamental regime, but as credit expands rapidly, the probability of being in the bubbly regime rises. By mid-2005, the bubble is becoming more likely, with a smoothed probability above 50%. Between 2007 and early 2009, our exercise reveals that the bubble was in full swing. Importantly, growth is bubble-driven in this period, which is an interesting contrast to the productivity-driven growth in the 1990s. At its peak, credit in the data is explained by a combination of bubbles and a favorable productivity shock. The bubble disappears in the early 2010s.

During the initial phase of the Great Recession, credit is in correction territory but still high compared to the 1990s. As a consequence, our approach identifies this stage of the crisis as the result of a sharp decline in investment demand due to an exogenous shock to preferences. But as the contraction in credit continued and the economy grew at lackluster rates, the fundamental regime becomes more likely, to the point where it is the prevalent regime since 2011. It is worth noting that our estimate of the bubbly episode lasts longer than other researchers have found (Jorda et al., 2015). This is due to the evolution of aggregate credit, peaking at the end of 2008 and slowly retrenching afterward, the latter of which Ivashina and Scharfstein (2010) attribute to the extensive use of existing lines of credit during 2009 and 2010. Ideally, we would use newly issued credit rather than total credit to better capture the narrative behind the crisis. However, to the best of our knowledge, such data are not available at the frequency and length required for our purpose.

It is worth noticing that a similar bubble regime would emerge if we used the Case-Shiller house price index. The main difference is that the bubble would collapse around the first quarter of 2008. The reason is that the credit-to-GDP ratio's dynamic tracks closely that of the Shiller-Case-to-GDP ratio except for the early collapse of the housing index.



Figure 14: Probability of Bubbly Regime

For the financial constraints of $\phi = 0.19$ considered in the main text, the average growth rates and credit-to-GDP are off the values in the data seen during the bubbly episode in the 2000s. One possibility, used in the paper, is to introduce a constant and estimate it to offset the difference. Alternatively, one can change the financial constraints to match the average growth rate during the presumptive bubbly period, with the caveat that we impose the dates when the bubble exists a priori. Figure 15 shows the estimated path of the probability of the fundamental (upper panel) and bubbly (lower panel) regimes under this specification. Clearly, the paths are consistent with those reported in the paper.

In the main text, we estimate the regimes using the sample 1984.Q1-2017.Q4. One can extend the sample to include the pre-Great Moderation era 1960.Q1-1983.Q4 but this brings a complication. Growth was strong during that period and credit-to-GDP was above average. Through the lens of our benchmark model, this points to a bubble. However, most economic observers would agree that there was no bubble during those years. To cope with this issue, we add a third regime that allows for high growth and average credit. Figure 16 shows the probabilities of each regime from this alternative model. As one can see, the main message remains. The high growth/high credit of the 2000s was most likely associated with the occurrence of a bubble in the economy. We also see that the economy spent most of the 1960s and 1970s in the third regime.



Figure 15: Regime Probabilities with Tighter Liquidity



Figure 16: Regime Probabilities Extended Sample